RESILIENCE TO EARLY LIFE SHOCKS

Evidence from the Interaction of a Randomized Controlled Trial and a Natural Experiment

Snaebjorn Gunnsteinsson Achyuta Adhvaryu[†] Parul Christian Alain Labrique Jonathan Sugimoto Abu Ahmed Shamim Keith P. West, Jr.

July 4, 2014

Abstract

Recent studies have documented the immediate and lasting effects of trauma in early life, but little is understood about how to protect children from these negative impacts. Protective effects are difficult to identify empirically, because both preventative and corrective investments are endogenous choices. We leverage a unique combination of events, in which a tornado struck an area of north-west Bangladesh involved in a double-blind cluster randomized controlled trial (RCT) of vitamin A supplementation—along with detailed birth records, morbidity and anthropometric measurements at 0-6 months—to test whether early life investment can mitigate the negative effects of experiencing a natural disaster. Exposure to the tornado *in utero* and at 0-3 months had large negative impacts on infants' anthropometric outcomes, and increased the frequency of severe fevers. However, infants treated with vitamin A at birth through the RCT were largely protected from these effects. Our results imply that simple health interventions can protect effectively against trauma in early life and that more research on the role of micronutrients in infant's resilience to shocks is likely to be valuable.

JEL: 112, 115, 118, J13, Q54, O12 Keywords: early life, child health, resilience, natural disasters, micronutrient supplementation

^{*}Special thanks to Prashant Bharadwaj, Paul Gertler, Jess Goldberg, Pam Jakiela, Anant Nyshadham, Rafael Perez-Escamilla, John Strauss, Atheen Venkataramani, and seminar participants at Maryland, USC, the CDC, and the NBER Children's meeting for helpful discussions. Adhvaryu gratefully acknowledges funding from the NIH/NICHD (5K01HD071949).

[†]Corresponding author. adhvaryu@umich.edu. 701 Tappan Street, Ann Arbor, MI 48109.

1 Introduction

Recent studies from around the world have demonstrated that trauma *in utero* and in early life can have large impacts on the health and survival of infants and young children. The sources of trauma are many and variegated. Air and water pollution (Chay and Greenstone, 2003a,b; Currie and Neidell, 2005; Currie and Walker, 2011; Currie et al., 2009, 2011; Greenstone and Hanna, 2011); natural disasters (Cas et al., 2014; Frankenberg et al., 2011); income shocks (Baird et al., 2011; Bhalotra, 2010); nutrient scarcity (Almond et al., 2011); maternal stress (Persson and Rossin-Slater, 2014); poor sanitation (Watson, 2006); and limited access health care (Almond et al., 2006) all have adverse effects on child health and wellbeing.

This mounting evidence begs the question: how do we protect vulnerable children? In other words, is it possible to engender resilience to early life shocks? This question is not answered easily. While it is plausible that exposure to many types of early-life shocks, including those mentioned above, is effectively random, measures taken to prevent negative impacts—and measures to mitigate impacts once a shock has occurred—are likely not random at all. They are deliberate choices. The extent to which parents invest in restoring their child's wellbeing after a shock is likely correlated with unobserved characteristics that also determine the child's outcomes. Assessing outcomes after these endogenous choices thus cannot produce a rigorous conclusion regarding how much a particular investment actually contributes to resilience.

In this paper, we study a unique situation that by chance combined an exogenous negative shock with a randomized health intervention in early life, thus allowing us to answer this question (applied, of course, to our particular setting). On March 20, 2005, a tornado struck several areas of northwest Bangladesh that were involved in a double-blind cluster randomized controlled trial (RCT) of maternal and newborn vitamin A supplementation. The tornado killed 56 people and injured almost 4000, and generated significant property damage in about 7 percent of the villages under study (Sugimoto et al., 2011). Both treatment and control villages were affected. We leverage this rare combination of events–along with detailed birth records, morbidity and anthropometric measurements at 0, 3, and 6 months–to test whether vitamin A supplementation mitigates the negative effects of experiencing a natural disaster in early life.¹

¹The protective role of vitamin A has a well charted physiological basis. Vitamin A promotes the functioning of neutrophils, macrophages, and natural killer cells – vital components of the body's immune system. It also helps restore innate immunity after infection by promoting the normal regeneration of mucosal barriers (Stephensen, 2001).

Our main focus is on the role of infant supplementation in promoting resilience. This is for two reasons. First, the primary goal of the prenatal supplementation RCT was to evaluate vitamin A's ability to stem maternal mortality, while the nested newborn supplementation trial focused directly on infant mortality. Second, the maternal trial showed small and statistically insignificant impacts of prenatal vitamin A or β -carotene supplementation on maternal, fetal, and infant mortality. The infant trial, on the other hand, found very large impacts on infant mortality–approximately a 15% reduction compared to the placebo group. For both reasons, it is more probable that infant supplementation exhibited a larger protective capacity than prenatal vitamin A. Consistent with this interpretation, we find little impact of the maternal trial on birth outcomes, anthropometry in early life, and the like, and little evidence that it is protective, while infant supplementation exhibits large protective effects.

Our empirical strategy exploits the contemporaneous combination of this RCT and a natural experiment. To estimate the effects of the tornado, we compare the health of cohorts of infants who were *in utero* or in early life (0-3 months) at the time of the tornado to earlier and later cohorts, across villages (also called sectors) falling within and outside the tornado's path. Then, to identify the potential protective effects of vitamin A supplementation, we add a third difference, across treatment and control sectors.

We show that exposure to the tornado *in utero* and at 0-3 months had substantial negative impacts on key birth outcomes and anthropometric measures in early infancy–namely mid-upper arm circumference (MUAC) and chest circumference (CC)–and increased the frequency of severe fevers in infancy. However, those treated with vitamin A at birth through the RCT were effectively protected from these deleterious effects. For example, a standardized anthropometric index at 3 months, constructed for mean effects analysis (Kling et al., 2007), dropped by .31 SD if the child was exposed to the tornado at 0-3 months in a placebo (control) sector. That difference all but vanishes for the same exposed cohort in treatment sectors. This pattern is consistent for anthropometrics at 6 months. Results on the incidence of fevers between 0 and 3 months reinforce the results on anthropometric outcomes: exposure to the tornado in the first three months of life increases the incidence of severe fevers, but this impact is wholly blunted in treatment sectors.

To our knowledge, ours is the first study to demonstrate in rigorous fashion that it is possible to protect against the deleterious effects of early life trauma. We show that health investments in early life can engender resilience to negative shocks. This result adds to our understanding of the health and economic impacts of changes in the early-life environment (Almond and Currie, 2011; Currie and Vogl, 2012; Heckman, 2006, 2007). Crucial to this understanding is the interplay between endowments and investments in the production of child quality (Almond and Mazumder, 2012; Bhalotra and Venkataramani, 2011; Cunha et al., 2013, 2010). We show that, at least in infancy, there is a high degree of substitutability between endowments and investments. In particular, vitamin A supplementation can make up for deficits in health endowments generated by tornado exposure. This amounts to a confirmation of the structural estimates of Cunha et al. (2010), using for the first time independent, exogenous sources of variation in both the endowment and investment. Since the RCT did not track infants beyond 6 months, we cannot determine whether this substitutability persists for longer-term outcomes. Though much more work is needed in this area, our findings suggest that it is indeed possible to protect young children from the lifelong disadvantage that can result from early life trauma.

Our result is also very related to recent work suggesting that intervention at birth (Almond et al., 2011, 2010, 2006; Bharadwaj et al., 2013) and investment in children (Gould et al., 2011; Kling et al., 2007) can in some cases correct for health-related or economic disadvantage. These studies show that early investments can improve the survival, health, and general welfare of children with low baseline health or economic status.

Our findings add to this literature in two ways. First, we measure a preventative–as opposed to a corrective–effect of investments at birth. Taken together, then, the evidence to date demonstrates that investments at birth work toward both improving outcomes after fetal disadvantage as well as protecting against trauma in early infancy. Second, this previous work has focused on identifying the returns to early intervention for children with poor endowments. But what would the returns have been if we had provided the same level of investment to less at-risk children? A large part of this gap in the literature is because the policies evaluated explicitly target at-risk children. We leverage the unique combination of an RCT and a natural experiment to quantify the impacts of intervention across both "disadvantaged" (exposed to the tornado) and "healthy" (not exposed) children. We show that that the returns to vitamin A supplementation are much larger for disadvantaged infants compared to those for healthier babies.

Finally, we provide evidence in support of policies encouraging vitamin A supplementation at birth in low-income contexts. Our results suggest that much of the impact of supplementation can be attributed to the large benefits accruing to the most distressed infants (in this case, to tornado-affected infants, and more suggestively, to low birth weight babies). Moreover, the effect of infant vitamin A supplementation on anthropometric measures at 3 and 6 months for "healthy" children (not exposed to the tornado) is very small. This idea–that there exists significant heterogeneity in the impact of basic supplementation–has not been properly emphasized in the public health literature on vitamin trials in low-income contexts.² To enhance their impact, supplementation policies should thus target distressed infants, particularly those living through traumatic experiences – natural disasters, disease outbreaks, war, and the like – in the first few months of life.

The remainder of the paper is organized as follows. Section 2 provides details regarding the vitamin A supplementation RCT and the tornado event. Section 3 describes our data, and section 4 describes the empirical strategy. Section 5 describes the results, and section 6 concludes.

2 Context

2.1 The RCT

The two randomized field experiments we study were nested double-blind placebo-controlled cluster randomized trials of maternal and newborn vitamin A (and in the maternal trial, β -carotene) supplementation in Bangladesh, conducted from 2001 to 2007. These trials and the tornado survey referred to below were all approved by the Institutional Review Board of the Bloomberg School of Public Health, Johns Hopkins University, and the Ethics Committee of the Bangladesh Medical Research Council. Each of the trials was pre-registered at clinicaltrials.gov; Identifiers: NCT00198822 (maternal trial) and NCT00128557 (infant trial).

The RCTs are part of the JiVitA Bangladesh international nutrition research project on maternal and child health. Both trials were conducted in a contiguous 435 square kilometer area in northwest Bangladesh, in Rangpur Division, with an estimated population of about 600,000. The study site is typical of rural Bangladesh, lying at approximately the 35th percentile of the distribution of economic and quality of life indicators among rural areas in Bangladesh (see Labrique et al. (2011) for more details on representativeness of the study area).

The study area was subdivided into 596 sectors, each of which was populated with 107 to 377 households at baseline. These sectors were randomized using a 3x2 cluster randomized factorial design with three different groups for pregnant women and 2 groups for their newborn children. The 3-group ran-

²See, e.g., Binka et al. (1995); Grotto et al. (2003); Imdad et al. (2011); Zeba et al. (2008).

domization (maternal trial) used a geographic block randomization, which is described in detail in West et al. (2011). The 2-group randomization (infant trial) was done by geographic block randomization, where each block was defined within one of the three earlier groups, as described in Klemm et al. (2008).

All married women in the study area in 2001 (totaling 102,769) and newlywed women (during the study, totaling 27,711), ages 13-45, were surveilled for pregnancy. In total, 60,294 pregnancies were identified and, if consent was given (>99% of cases), the pregnant woman was enrolled in the maternal supplementation study. The infant trial was nested within the maternal trial and was conducted during part of the maternal trial, or between January 2004 and December 2006. A total of 15,937 infants received supplementation (or placebo) directly at birth or shortly thereafter (79% within 24 hours; 90% within 1 week) and were followed until 6 months after birth.

The two treatment groups in the maternal trial received the recommended weekly allowance of vitamin A, either in the form of vitamin A or β -carotene (which the body converts into vitamin A), as weekly supplements from first trimester through 12 weeks post-partum, while the control group received a placebo supplement. Live-born infants in each sector were randomized to receive either 50,000 International Units (IU) of vitamin A once at birth or to receive placebo once at the same time. Further information on field procedures and other details can be found in Labrique et al. (2011), West et al. (2011) and Klemm et al. (2008).

As mentioned in the introduction, our analysis focuses primarily on the infant RCT. The main purpose of the maternal trial was the assess the impact of prenatal supplementation on maternal outcomes, which should be, of course, indirectly linked to infant resilience, but not in the direct way that the infant RCT was. Moreover, vitamin A treatment at birth generated large impacts on infant mortality, while maternal supplementation did not; we verify that these contrasting sets of results apply for anthropometric outcomes as well.

2.2 The Tornado

On the night of March 20th, 2005, a tornado swept through Gaibandha District, affecting about 7% of the study area (Sugimoto et al., 2011) (see Figure 1). Between August and October 2005 each household in affected areas was visited by a survey enumerator, who asked questions on mortality and morbidity of household members as well as damage to homes as a result of the tornado. Based on this survey, the tornado resulted in 56 deaths, injured 3,710 people and destroyed 3,540 houses (Sugimoto et al.,

Figure 1: Area damaged by the tornado. The figure was produced by the JiVitA GIS Unit.

2011). Out of 596 study sectors, 41 sector had some houses destroyed and in 24 sectors more than 20% of houses were destroyed. Many actors responded to the disaster – including local NGO's, local and national government and even the medical team of the two Vitamin A trials – providing medical care and supplies of rice, cash and building supplies. Our estimates below of the impact of the tornado should be considered as net of the effect that this relief effort had. Our evidence suggests that the tornado had no effect on the timing of supplementation or anthropometric and survey measurements. For instance, among infants in their second or third trimesters in-utero during the tornado those in the tornado area were supplemented within 24 hours at the rate of 80% while those outside of this area were dosed at the rate of 79.2%. Birth anthropometry for this same population was obtain within 7 days in the tornado area at the rate of 88.2% and outside this area at the rate of 89.5%.

3 Data

3.1 Sample

We include in the sample all infants in the trial born before July 20, 2006 (following Klemm et al. (2008)) for which informed consent was obtained but exclude non-singleton births (298 infants) and those we can not assign to a cohort due to missing data (63 infants). This leaves us with a final core analysis sample of 18,787 infants.

3.2 Summary Statistics

Table 1 reports means and standard deviations of important outcomes and control variables. We code as missing birth measures taken after 7 days and 3 and 6 month measures taken more than 8 weeks after the target date (in our regressions we also control for the date of measurement). We report means for the whole sample, as well as within and outside of tornado-affected sectors, and across treatment and control sectors within the tornado area. We also report differences in means across these sub-samples. Differences noted with asterisks denote statistically significant differences.

Babies in this area of Bangladesh are small relative to reference populations. The mean weight is 2.5 kg, exactly at the threshold for classification as low birth weight. Average length at birth in cm is approximately 46.7, a full 3 cm less than the reference US population. Head circumference is 32.7 cm at birth, which is 3 cm less than the same measurement for the reference US population. This difference (with respect to the reference population) shrinks slightly by 6 months: head circumference at 6 months is 40.89 cm as compared to 43.5 cm for reference infants.³

3.2.1 Comparisons across affected and unaffected areas and across study arms within tornado sectors

Means of health outcomes at birth and at 3 and 6 months are balanced across the tornado and nontornado areas for pre-tornado cohorts. There is some evidence that infants in tornado-affected sectors were slightly healthier, particularly by 6 months: 9 out of the 11 anthropometric measurements recorded are larger in the tornado area; 3 of these differences–mid-upper arm circumference, chest circumference, and an anthropometric index, all at 6 months–are statistically significant, though the differences are small in magnitude.⁴

Next, we compare means across treatment sectors (infants who received vitamin A supplementation at birth) and control (placebo) sectors within the tornado-affected area. Reassuringly, child outcomes (weight and anthropometry at 0, 3, and 6 months) are balanced across the treatment arms within the

³Data for reference populations are from the Centers for Disease Control and Prevention Growth Charts for the United States (Kuczmarski et al., 2000).

⁴The anthropometric index (AI) is constructed as the normalized sum of the three (normalized) independent measures of anthropometry we have in the data: mid-upper arm circumference (MUAC), head circumference (HC), and chest circumference (CC). The normalization centers the mean of the index over the whole sample at 0 and standardizes the variance to 1. The AI is meant to be a summary measure of anthropometry at 3 and 6 months. Also note that all difference statistics reported in this table are not clustered.

tornado area. Two maternal characteristics, parity and education, are different across vitamin A and placebo groups.

3.2.2 Dosing

Infants were dosed within hours of birth with either treatment (vitamin A) or placebo. The trial was double blind, so the implementation teams did not know whether they were dosing infants with treatment or placebo. In Table 2, we report dummies reflecting the distribution of time to dosing (in hours after birth) across key groups.

48 percent of infants were dosed within 6 hours of birth. 62 percent were dosed within 12 hours, and 74 percent by 24 hours. The dose timing distribution has a long right tail: 16 percent of infants were dosed more than 7 days after birth. Overall, the distribution of dosing timing was fairly similar across tornado and non-tornado sectors, and within the tornado area across vitamin A and placebo groups. We do see small significant differences in dosing less than 6 and 12 hours (note that this is prior to the tornado). Tornado areas were dosed slightly *earlier* than non-tornado areas.

In Table A5, we verify that for *exposed* infant cohorts, dosing was actually more equal across tornado and non-tornado areas, and, within the tornado area, across vitamin A and placebo sectors.

3.2.3 Attrition

Finally, we address the possibility that the tornado (or infant vitamin A treatment) resulted in differential attrition from the sample. If this were the case, selective attrition might bias the estimates of the impacts of early life trauma and resilience due to vitamin A exposure. Attrition here is treated as "all-cause," meaning that our main variable of interest is whether child outcomes were not observed, for whatever reason, during the 0, 3, and 6 month enumerator visits. The two specific causes of attrition are death during infancy and migration from the survey areas; the rest of attrition comes from unspecified reasons.

We assess the extent to which attrition is differential across tornado-affected and unaffected cohorts, and then assess whether the triple interaction with vitamin A treatment predicts differential attrition, as well. In our main analysis, we calculate *in utero* exposure based on the period between best-guess date of conception and date of birth, but of course in this analysis we do not have date of birth for babies who were not found for birth measurement. We thus calculate periods based on the last-known menstrual period (LMP), which is used in the calculation of best-guess date of conception as well. We define the *in utero* period as 0-270 days after LMP; then 0-3 month exposure as 271-360 days after LMP; and 4-6 month exposure as 361-450 days after LMP. In this way we can define periods even for babies who are no longer present in the sample, making attrition analysis possible.

The results are reported in Table 3. The odd columns in the table (1, 3, and 5) report results using a binary variable equal to 1 when the measure at a particular period (birth, 3 and 6 months) is missing. The even columns (2, 4, and 6) report a similar dummy but include babies for whom the the measure was very late (after 7 days for birth anthropometry, and 8 weeks after the target date for 3 and 6 month measurements). We assign these variables to missing in our analyses as they represent mismeasurement of outcomes that is not necessarily linear (i.e., would not necessarily be appropriately absorbed by including controls for age at measurement, which we do in all regressions). For a breakdown of conditional means in these variables across the groups of interest, please refer to Table A1 in the appendix.

Overall, results at birth and 3 months show very little evidence of differential attrition for tornadoexposed infants, and the point estimates on the triple interaction with infant supplementation treatment are also insignificantly different from 0. At 6 months, we see significant differential attrition for tornadoexposed infants, but no such differences in the triple interaction. That is, by 6 months, there appears to be more attrition for tornado-exposed infants. This may be because mortality effects kick in only after 3 months, or because the tornado's economic effects do not manifest immediately. Regardless, sample selection will be an issue at 6 months. Encouragingly, the attrition issue is not present at 3 months, and we largely find the same effects on anthropometry, both for the effects of the tornado and the resilience generated by vitamin A, in both periods.

4 Empirical Strategy

4.1 Sources of variation

We leverage three sources of variation to identify the protective effect of vitamin A: 1) spatial variation in tornado exposure; 2) temporal overlap between the tornado event and key early life periods; and 3) the randomized allocation of vitamin A to newborns.

With regard to spatial variation, we compare infants born in sectors that were in the tornado's path with those born in sectors outside this path. Our baseline definition of spatial exposure classifies a particular sector as exposed if there was any tornado damage in the sector. Under this definition, 41 sectors, or 7 percent of all sectors involved in the RCT, were exposed. Since several sectors experienced a low degree of damage (less than 20 percent of homes were destroyed in 17 sectors), we may be misclassifying these sectors as "exposed" according to the baseline definition. We can thus define an alternative that includes only sectors affected to a greater degree – for example, sectors in which greater than 20 percent of homes were destroyed. Results are qualitatively unchanged when we use the alternative definition of spatial exposure. The caveat to using this alternative is that as we increase the cutoff, the number of exposed sectors shrinks, leading to imbalance in the sample across the exposed and unexposed groups, thus power goes down. A summary of tornado-affected sectors is provided in Table 4.

Second, we construct dummies for two main time periods of early exposure: the prenatal period (i.e., the infant was *in utero* during the tornado event) and early life (i.e., the infant was either 0-10 weeks or 0-22 during the tornado event, depending on the timing of the measurement of the outcome variable). Throughout the paper we define the *in utero* period as the time between our best guess of the date of conception (based on the last menstrual period) and birth. This definition has the advantage of being the normal definition of the *in utero* period but the disadvantage that it induces a mechanical correlation between exposure to shocks *in utero* and gestational length (Currie and Rossin-Slater, 2013). To account for this, control for best-guess length of gestation in our main specification.

For the postpartum period, we chose 10 rather than 12 (and 22 rather than 24) weeks because our measurements of anthropometric outcomes and the survey of mothers were done at 12 and 24 weeks postpartum. We thus leave at least 2 weeks' gap for the effects of the tornado to manifest themselves in infant outcomes.⁵

Third, we use randomized variation in the allocation of vitamin A to newborns by sector. Accordingly, we construct a dummy for whether the infant was born in a treatment sector, meaning he was dosed with vitamin A at birth. As explained earlier, supplementation at birth in the RCT was crossrandomized with prenatal supplementation. We focus on supplementation at birth in the body of the paper, and describe the prenatal supplementation results in the appendix.

⁵For regressions using the history of fever episodes at 12 and 24 weeks as the dependent variables, we use the full 12 and 24 weeks of exposure, respectively, since fever should respond more immediately than anthropometry.

4.2 Estimation

Perhaps the most intuitive candidate strategy for identification of a protective effect for vitamin A is to compare infants born in some window of time around the tornado event, inside and outside the tornado area, and across treatment (vitamin A supplementation) versus control (placebo) sectors–a difference in differences strategy. This would, however, require two fairly strong assumptions for identification. First, we would have to assume that the tornado hit a random subset of sectors. This is clearly violated, as the tornado affected a spatially contiguous area. Second, we would need that the randomization was balanced both inside and outside the tornado-affected area. Due to the small number of sectors inside the tornado area, this assumption may not hold.

Our strategy, which leverages the timing of births using a triple difference specification, allows for much weaker assumptions. In particular, in the difference in differences strategy above, we would rely on a single (spatial) difference across tornado and non-tornado areas to identify the effects of the tornado. In contrast, here we can rely on both spatial and temporal variation (around the tornado window) to identify the tornado effect. That is, we can compare babies born at different times (within and outside of a window around the tornado event), across sectors affected by and unaffected by the tornado. Then, we can essentially compare the size of this effect across vitamin A treatment sectors and placebo sectors. This strategy also lowers concern about imbalance between the vitamin A and placebo sectors as we estimate how birth outcomes change over time rather than comparing main effects across sectors.

Finally, we must also assume that the tornado hit vitamin A and placebo sectors equally hard. This can be checked in the data. In fact, the average number of houses destroyed in the tornado hit vitamin A sectors was 33.7% compared to 47.6% in the tornado hit control sections. This is in part because the 6 sectors hit hardest (by this measure) were all control sectors. Excluding those sectors the average damage in tornado hit vitamin A sectors is 33.5% compared to 28.7% in the tornado hit control sectors. However, as discussed below and displayed in Table 15, our findings are unchanged (and in fact statistically stronger) if we exclude those sectors.

We thus estimate a triple difference across the three dimensions described above to identify the protective effect of vitamin A. We assess the impact of the tornado by comparing outcomes for infants across sectors affected by the tornado v. unaffected sectors and for those whose prenatal and early life periods coincided with the tornado timing v. those for whom these periods did not. We then take

a third difference across treatment v. control sectors, to estimate the protective effect of vitamin A supplementation at birth.

In section 5, we begin by presenting the raw conditional means of key outcome variables (anthropometry, fever incidence, and mortality) within these groups. The differences across groups suggest a large negative impact of tornado exposure and an equally large protective effect of vitamin A supplementation.

We then estimate the full triple difference specification via ordinary least squares (OLS):

$$O_{ij} = \alpha + VitA_j \left(\beta_1 U_i + \beta_2 E_i + \beta_3 T_j U_i + \beta_4 T_j E_i\right)$$

$$+ \gamma_1 U_i + \gamma_2 E_i + \gamma_3 T_j U_i + \gamma_4 T_j E_i + X'_{ij} \delta + \mu_j + \varepsilon_{ij}.$$

$$(1)$$

Here, *i* denotes infant and *j* denotes sector. O_{ij} is a health outcome measure. T_j is a dummy for tornado-exposed sector. U_i is a dummy that is 1 if the infant was *in utero* during the tornado event, and E_i is a dummy that is 1 if the infant was 0-10 weeks during the tornado. $VitA_j$ is a dummy for treatment sector in the RCT. μ_j is a sector fixed effect, which absorbs the main effects of vitamin A treatment and tornado sector classification (as well as the interaction of those two). X_{ij} is a vector of determinants of child health used as controls. Throughout the analysis, we cluster standard errors at the sector level, but in Table 13 we replicate the main results using two-way clustering on sector and week of birth.

5 Results

5.1 Birth outcomes

We begin with impacts on birth outcomes. The main purposes of this analysis are 1) to estimate the impacts of *in utero* exposure to the tornado on birth outcomes, and 2) to assess the protective effect of maternal (prenatal) supplementation with vitamin A or β -carotene. The main outcomes we assess as birth are birth weight, gestational age, small for gestational age (below 25th percentile birth weight adjusted for gestational age), height, MUAC, HC, CC, and a mean effects index comprised of all of the measurements at birth just mentioned.

Table 5 reports results of regressions of these outcomes on tornado exposure and its interaction with prenatal vitamin A supplementation. For nearly all variables, we estimate a substantial negative

impact of *in utero* exposure. For example, the likelihood of low birth weight (birth weight < 2.5 kg) is higher by 15 percentage points; height decreases by .54 cm; and other anthropometric measurements (MUAC, HC, and CC) decline by .25 to .47 cm. In short, we find strong evidence of negative impacts of *in utero* tornado exposure on outcomes at birth.

The coefficients on the interaction of exposure with maternal supplementation with vitamin A are in general imprecisely estimated, but show a weak pattern towards protection. For example, the likelihood of low birth weight increases by 15 points with *in utero* tornado exposure, but one-third of that increase disappears for mothers receiving vitamin A supplementation during pregnancy. Overall, however, the results are not precise enough to claim conclusively that prenatal vitamin A or β -carotene supplementation has a role in protecting against *in utero* assault.

5.2 Anthropometry at 3 and 6 months

5.2.1 Preliminary Evidence

Next we present preliminary evidence from means across groups of the anthropometric index (AI) at 3 months. AI group means are reported in Table 6. There are three panels of two-by-two tables; each panel corresponds to a time period of birth relative to the time of the tornado event. The first panel reports means for infants born outside the tornado window, the second reports means for those with *in utero* exposure, and the third reports means for those with early life exposure. Each panel reports means across tornado v. non-tornado sectors and across vitamin A treatment v. placebo sectors (where treatment is supplementation with a single vitamin A dose at birth). This generates 4 means per panel. We also calculate differences across these groups, and a difference in differences estimate within each panel. Finally, we take the triple difference across panels. We report two such triple differences – one comparing *in utero* exposure (second panel) to no exposure (first panel), and one comparing early life exposure (third panel) to no exposure.

We begin with the first panel of Table 6. We see that outside the tornado window, there are fairly large differences across sectors falling within and outside of the tornado's path. In particular, babies born in tornado sectors have significantly better anthropometry (by .1 SD in treatment sectors and .14 SD in control sectors), in both vitamin A treatment and placebo sectors. This fact underscores the non-random selection of tornado sectors, and therefore the need for a third difference over time.

Reassuringly, there are no significant differences across treatment and placebo sectors, though, as we might expect given the smaller sample, the estimates are somewhat imprecise in tornado areas. The difference in differences estimate is also not significantly different from 0.

The second panel of Table 6 reports AI means at 3 months for infants exposed to the tornado *in utero*. In placebo sectors, mean AI in tornado areas was the same as in non-tornado areas (as compared to .14 SD higher in tornado areas for infants outside the tornado window), suggesting that the tornado had a large adverse effect on infants who were exposed *in utero*. In vitamin A treatment sectors, this difference was .1 SD, and thus the difference in differences estimate is essentially 0. None of these estimates is significantly different from 0. Overall, this panel provides limited evidence in support of the hypotheses that the tornado had an adverse impact on babies exposed *in utero*, and that vitamin A had a protective effect.

The third panel of Table 6, with 3 month AI means for infant exposed in early life to the tornado, shows large differences across groups. Non-tornado area means are essentially 0, as we would expect. In tornado areas, however, the mean AI in placebo sectors is -.11 SD, compared to .15 SD for infants outside the tornado window. This difference suggests a large negative impact of the tornado for infants exposed in the first 10 weeks of life. Yet in vitamin A treatment sectors, the tornado sector mean for exposed infants is the same as the mean AI in tornado areas (across vitamin A and placebo sectors) outside of the tornado time window.

Below these three panels, we report triple difference estimates for *in utero* and early life exposure. These estimates capture the corrective and protective effects of vitamin A, respectively. The triple difference estimate for the corrective effect of vitamin A after *in utero* exposure is .14 (and insignificant), and the estimate for the preventative effect is .31 and significantly different from 0.

We report an analogous set of means for AI at 6 months in Table 7. The pattern of differences is exactly the same as those reported in the previous table. The first panel demonstrates that, outside the tornado time window, infants in tornado sectors are significantly larger at 6 months than those born in non-tornado sectors. From the means in the second panel for infants exposed *in utero*, we see the same pattern, suggesting that the tornado did not have a large impact by 6 months for those infants exposed during gestation.

The third panel, however, with AI means for infants exposed in the first 22 weeks of life, suggests that the tornado did indeed impact these infants-the difference in the placebo group across tornado and non-

tornado sectors is -.04, compared to .17 for infants outside the tornado time window, indicating a large impact of tornado exposure on these infants. Yet in the vitamin A treatment group, the difference across tornado and non-tornado sectors is .20, *larger* than the difference outside of the tornado time window. That is, those infants born in the vitamin A sectors appear protected from the negative impacts of the tornado in early life. The triple difference estimates are consistent with the difference in differences estimates from the separate panels and are significantly different from 0.

5.2.2 Regression Results

Next, we estimate impacts of the tornado for infants exposed *in utero* and in early life, as well as the corrective and protective effects of vitamin A supplementation, using linear regression models of the form shown in equation 1. We regress AI and, separately, its individual components–MUAC, CC, and HC at 3 and 6 months⁶–on the triple interaction of a vitamin A treatment sector dummy, a tornado area dummy, and a dummy for either *in utero* or early life (0-10 weeks) exposure (we include both triple interactions in the specification). All lower-level interactions and main effects are also included, as well as sector fixed effects, which absorb some of these main effects. In addition, we control in all models for the best guess length of gestation in weeks; in some models we also control for maternal MUAC, maternal height, and a living standards index, which is generated via principal components analysis using household assets data.

The results for AI and its components at 3 months are reported in Table 8. The triple interactions for *in utero* and early life exposure are reported at the top of the table. These tell us the magnitudes of the corrective and protective effects of vitamin A, respectively. Next, we report the double interactions with tornado area and being born in the tornado time window. These coefficients tell us the size of the negative effects of the tornado. We then report other double interactions, main effects, and controls.

The results in Table 8 generally confirm what we learned from the preliminary evidence from AI group means. We discuss AI results first, reported in columns 1 and 2 of the table. The interaction of birth in tornado area with early life exposure has a large and negative coefficient (-.31, p<.05); the coefficient on the interaction with in utero exposure is also negative, but smaller (-.17) and insignificant. In other words, we find that by 3 months, infants exposed in early life to the tornado were more than

⁶Note that mid-upper arm circumference (MUAC), chest-circumference (CC) and head circumference (HC) are all measured in centimeters.

.3 SD smaller than non-exposed infants. The negative impacts of *in utero* exposure appear to diminish by 3 months, though the coefficient estimate is still negative.

The triple interaction coefficients suggest that vitamin A has a significant protective effect, dampening the impact of tornado exposure substantially. The coefficient on the triple interaction with early life exposure is .41 without controls (column 1) and .32 with controls (column 2), indicating that infants in vitamin A sectors were not significantly affected by the tornado, while those in placebo sectors experienced large negative impacts. The patterns for MUAC, CC and HC are similar; triple difference estimates of the protective effect of vitamin A are significantly different from 0 in both specifications (with and without controls) for MUAC and CC, but not HC. Length of gestation, maternal MUAC and height, as well as living standards, are all positively associated with anthropometry.

In Table 9, we show the negative impacts of the tornado and the protective effect of vitamin A both continue to be salient for anthropometry at 6 months. Indeed, the estimates of the tornado impact on AI, MUAC, CC and HC are all larger at 6 months compared to 3 months, as are the protective effects (read off the triple interactions). Again, we only find significant effects for early life exposure; in particular, only for exposure in the first 3 months, but not at 4-6 months. The impacts of *in utero* tornado exposure and mitigative impacts are both insignificant.

5.3 Fever Episodes

Next we present evidence on fever episodes, one mechanism underlying the impacts on anthropometric outcomes. Since vitamin A's primary role in infancy is to strengthen the immune system, we would expect that infants dosed with vitamin A at birth are less prone to fevers occurring because of poor nutrition, sanitation, and the like following tornado exposure. We test this hypothesis using data on infant fever episodes reported by mothers at 3 and 6 months.

Table 10 shows group means for the number of high (as classified by the respondent) fever episodes experienced in the first 3 months of life. The first panel shows that there were no differences in fever episodes across tornado and non-tornado areas or across vitamin A and placebo areas for infants outside the tornado time window. The average infant had just under 1 "high" fever episode in the first 3 months of life.

In the second panel, we report the same group means for infants born in the *in utero* exposure time window relative to the tornado. Here, we see significant differences in number of fever episodes across

the tornado and non-tornado areas, indicating that *in utero* exposure was indeed detrimental for infants. In both vitamin A and placebo sectors, infants born in tornado areas had approximately .22 and .34 more fever episodes, respectively. The double difference estimate (.12) is small and insignificant, indicating that vitamin A had no corrective effect for this cohort of exposed infants. This evidence is consistent with the small observed impact on anthropometric outcomes.

In the third panel of Table 10, we report group means for cohorts coinciding with early life exposure to the tornado. We find that mean fever episodes in non-tornado areas was .86, consistent with the means for other cohorts reported in the previous panels. But the mean number of episodes in tornado areas that were placebo sectors in the RCT was 1.34, nearly .5 episodes (more than 50%) greater than non-tornado areas, while in tornado-exposed vitamin A sectors, the mean was unchanged, at .88. This difference is echoed in the triple difference estimates of the protective effect of vitamin A, offering powerful descriptive evidence that fevers indeed increased for infants exposed in early life, but this impact was mitigated by vitamin A supplementation.

The regression results for fever and episodes for 0-3 and 4-6 months, reported in Table 11, confirm these preliminary findings. The double interactions between tornado area and tornado time windows show that exposure to the natural disaster increased the incidence of fever for most types of exposure–*in utero* (in some specifications), at 0-3 months, and at 4-6 months. However, vitamin A had a protective effect for early life exposure (0-3 months), though this coefficient becomes insignificant when the triple interaction for 4-6 month exposure is added to the specification.

5.4 Robustness Checks

5.4.1 Restricting Control Group to Pre-Tornado Cohorts

In our main analysis, infants conceived after the tornado are considered part of the (temporal) control group. It is possible that these infants were affected by the aftermath of the tornado; for example, sanitation and health infrastructure likely took time to rebuild in affected areas, so infants born in some window well after the tornado could still have been exposed to its negative impacts.

To account for this possibility, we include additional interaction terms that effectively remove the cohort conceived after the tornado from the control group. Thus all cohorts are now compared only to the cohort born more than 3 months before the tornado. The results are reported in Table 12. We find

that the results on AI, MUAC, and fever all retain their magnitudes and statistical significance. As we would predict, the inclusion of the cohort conceived after the tornado makes the estimates of tornado impact in early life and the protective effect of vitamin A stronger.

5.4.2 Two-way Clustering

In Table 13, we replicate the main results on AI, MUAC, and fever at 3 and 6 months using twoway clustering of standard errors on cluster (the unit of randomization) and week of birth (the unit of variation in exposure). The precision of estimates of the negative impacts of tornado exposure and the protective effect of infant vitamin A supplementation is largely unchanged.

5.4.3 Changing the Definition of Tornado Exposure

Our baseline variable for tornado exposure is an indicator that equals 1 for each infant if the sector in which the infant was born had a positive percentage of homes destroyed. Exposure might be better defined using higher cutoffs, because low levels of tornado damage may not generate large enough impacts on infant anthropometry to detect statistically. On the other hand, since defining the exposure cutoff at 0 yields about 10 percent of sectors defined as exposed, increasing the cutoff will yield a very small fraction of the sample classified as exposed. This creates small cells of infants who were born in exposed sectors around the time of the tornado, and thus estimates become more imprecise. We favor the "any exposure" definition to mitigate this latter concern, but we check whether exposure at higher cutoffs has similar effects.

To do implement this, we take our basic specification and divide tornado exposure into two dummies – one for 0-20 percent of homes destroyed, and one for 20-100 percent. We look at the double and triple interaction coefficients for each of these indicators. The results are presented in Table 14. We find in general that the size of the coefficients for both the impact of the tornado and the protective effect of vitamin A are not statistically different for both definitions of exposure, suggesting that our baseline definition seems to capture the appropriate variation in exposure.

Another possible concern for our estimates is that 6 of the most damaged sectors were all control sectors. As a result, among the tornado hit sectors, 47.6% of houses in control sectors were destroyed compared to 33.7% of houses in vitamin A sectors. If we exclude those 6 sectors most damaged then the balance shifts such that 28.7% of houses in control sectors are destroyed compared to 33.5% in

the vitamin A sectors. We re-estimated our key regressions excluding these 6 sectors and report the findings in Table 15. Our findings are similar and in fact statistically stronger than before. It therefore appears unlikely that our findings are driven by control sectors being especially hard hit, at least not by our measure of residential house destruction.

6 Conclusion

Infants are vulnerable to a variety of assaults *in utero* and in early life. Quantifying the negative effects of environmental factors, income and nutritional scarcity, and natural disasters on infant health and survival is the focus of a rapidly expanding set of studies in economics. We know from this work that impacts, particularly in low-income contexts, can be large and long-lasting. But we have little rigorous empirical evidence that intervening in early life can change outcomes for children exposed to trauma.

In this study, we leverage the unique combination of a natural disaster during an RCT to estimate the negative impacts of tornado exposure on birth outcomes, and the protective effect of vitamin A supplementation at birth. We find significant impacts of the tornado on anthropometric outcomes at 3 and 6 months. But babies who received a one-time dose of vitamin A at birth did not experience the same drops in anthropometric measures. Results on the incidence of fever episodes in infancy reinforce these findings, lending some insight into the mechanism through which the protective capacity of vitamin A operates.

The results this study demonstrate, to our knowledge for the first time, that simple interventions at birth can protect effectively against trauma in early life. This is important because improving the health and survival of infants, particularly in low-income countries, is a primary goal for global health policy. Moreover, a growing literature in economics shows that in addition to these immediate impacts, early life assaults have far-reaching long run consequences. Disease (Almond, 2006; Bleakley, 2007, 2010; Cutler et al., 2010), natural disasters (Currie and Rossin-Slater, 2013), income shocks (Maccini and Yang, 2009), and war (Akresh et al., 2012) all leave lasting scars on health, human capital, and welfare that persist well into adulthood. The role of public policy in mitigating these impacts or protecting against them is widely recognized but poorly understood. In large part the dearth of rigorous evidence on policy levers is due to the difficulty in finding overlapping episodes of early life trauma and an orthogonal natural experiment that changed the incentives for investing in children.

Our study takes a step toward filling this gap. Our results demonstrate a strong protective effect of one-time vitamin A supplementation at birth. We interpret this protection as evidence that, at least in very early life, endowments (as proxied for by tornado exposure) and investments (vitamin A) are substitutes. Whether this remains true when outcomes are measured in later childhood and adulthood is an open question. Although our findings hold up to various robustness checks and are consistent across a diverse set of outcomes (anthropometry, fever incidence and mortality), their strength is somewhat limited by the relatively small share of infants in the study affected by the tornado. Our results hopefully offer a valuable start and suggest that more research on the role of micronutrient deficiencies in infants' resilience to shocks is likely to be very valuable.

7 Author Affiliations

- Gunnsteinsson: University of Maryland
- Adhvaryu: University of Michigan
- Christian, Labrique and West: Johns Hopkins Bloomberg School of Public Health
- Sugimoto: Fred Hutchinson Cancer Research Center
- Shamim: Jivita Bangladesh

References

- Akresh, Richard, Sonia Bhalotra, Marinella Leone, and Una Okonkwo Osili, "War and Stature: Growing Up during the Nigerian Civil War," *The American Economic Review*, 2012, *102* (3), 273–277.
- Almond, Douglas, "Is the 1918 Influenza pandemic over? Long-term effects of in utero Influenza exposure in the post-1940 US population," *Journal of Political Economy*, 2006, *114* (4), 672–712.
- _ and Bhashkar Mazumder, "Fetal origins and parental responses," 2012.
- and Janet Currie, "Killing me softly: The fetal origins hypothesis," The Journal of Economic Perspectives, 2011, 25 (3), 153–172.
- ____, Hilary W Hoynes, and Diane Whitmore Schanzenbach, "Inside the war on poverty: the impact of food stamps on birth outcomes," *The Review of Economics and Statistics*, 2011, *93* (2), 387–403.
- ____, Joseph J Doyle, Amanda E Kowalski, and Heidi Williams, "Estimating marginal returns to medical care: Evidence from at-risk newborns," *The quarterly journal of economics*, 2010, *125* (2), 591–634.
- _, Kenneth Y Chay, and Michael Greenstone, "Civil rights, the war on poverty, and black-white convergence in infant mortality in the rural South and Mississippi," 2006.
- Baird, Sarah, Jed Friedman, and Norbert Schady, "Aggregate income shocks and infant mortality in the developing world," *Review of Economics and Statistics*, 2011, *93* (3), 847–856.
- Bhalotra, Sonia, "Fatal fluctuations? Cyclicality in infant mortality in India," *Journal of Development Economics*, 2010, 93 (1), 7–19.
- Bhalotra, Sonia R and Atheendar Venkataramani, "The captain of the men of death and his shadow: Long-run impacts of early life pneumonia exposure," Technical Report, Discussion Paper series, Forschungsinstitut zur Zukunft der Arbeit 2011.
- Bharadwaj, Prashant, Katrine Vellesen Loken, and Christopher Neilson, "Early life health interventions and academic achievement," *The American Economic Review*, 2013, *103* (5), 1862–1891.
- Binka, Fred N, David A Ross, Saul S Morris, Betty R Kirkwood, Paul Arthur, Nicola Dollimore, John O Gyapong, and Peter G Smith, "Vitamin A supplementation and childhood malaria in northern Ghana.," *The American journal of clinical nutrition*, 1995, *61* (4), 853–859.

- Bleakley, Hoyt, "Disease and development: evidence from hookworm eradication in the American South," *The Quarterly Journal of Economics*, 2007, *122* (1), 73–117.
- ____, "Malaria Eradication in the Americas: A Retrospective Analysis of Childhood Exposure," American Economic Journal: Applied Economics, 2010, 2 (2), 1–45.
- Cas, Ava Gail, Elizabeth Frankenberg, Wayan Suriastini, and Duncan Thomas, "The impact of parental death on child well-being: Evidence from the Indian Ocean tsunami," *Demography*, 2014, *51* (2), 437– 457.
- Chay, Kenneth and Michael Greenstone, "Air Quality, Infant Mortality, and the Clean Air Act of 1970," NBER Working Paper, 2003, (w10053).
- Chay, Kenneth Y and Michael Greenstone, "The impact of air pollution on infant mortality: evidence from geographic variation in pollution shocks induced by a recession," *The quarterly journal of economics*, 2003, *118* (3), 1121–1167.
- Cunha, Flávio, Irma Elo, and Jennifer Culhane, "Eliciting maternal expectations about the technology of cognitive skill formation," Technical Report, National Bureau of Economic Research 2013.
- Cunha, Flavio, James J Heckman, and Susanne M Schennach, "Estimating the technology of cognitive and noncognitive skill formation," *Econometrica*, 2010, *78* (3), 883–931.
- Currie, Janet and Matthew Neidell, "Air pollution and infant health: What can we learn from California's recent experience?," *The Quarterly Journal of Economics*, 2005, *120* (3), 1003–1030.
- _ and Maya Rossin-Slater, "Weathering the storm: Hurricanes and birth outcomes," *Journal of health economics*, 2013, *32* (3), 487–503.
- and Reed Walker, "Traffic Congestion and Infant Health: Evidence from E-ZPass," American Economic Journal: Applied Economics, 2011, 3 (1), 65–90.
- and Tom Vogl, "Early-life health and adult circumstance in developing countries," Technical Report, National Bureau of Economic Research 2012.
- ____, Matthew Neidell, and Johannes F Schmieder, "Air pollution and infant health: Lessons from New Jersey," *Journal of health economics*, 2009, 28 (3), 688–703.

- ____, Michael Greenstone, and Enrico Moretti, "Superfund cleanups and infant health," Technical Report, National Bureau of Economic Research 2011.
- Cutler, David, Winnie Fung, Michael Kremer, Monica Singhal, and Tom Vogl, "Early life Malaria Exposure and Adult Outcomes: Evidence from Malaria Eradication in India," *American Economic Journal: Applied Economics*, 2010, *2* (2), 72–94.
- Frankenberg, Elizabeth, Thomas Gillespie, Samuel Preston, Bondan Sikoki, and Duncan Thomas, "Mortality, The Family and The Indian Ocean Tsunami," *The Economic Journal*, 2011, *121* (554), F162– F182.
- Gould, Eric D, Victor Lavy, and M Daniele Paserman, "Sixty years after the magic carpet ride: The long-run effect of the early childhood environment on social and economic outcomes," *The Review of Economic Studies*, 2011, *78* (3), 938–973.
- Greenstone, Michael and Rema Hanna, "Environmental regulations, air and water pollution, and infant mortality in India," Technical Report, National Bureau of Economic Research 2011.
- Grotto, Itamar, Marc Mimouni, Michael Gdalevich, and Daniel Mimouni, "Vitamin A supplementation and childhood morbidity from diarrhea and respiratory infections: a meta-analysis," *The Journal of pediatrics*, 2003, 142 (3), 297–304.
- Heckman, James J, "Skill formation and the economics of investing in disadvantaged children," *Science*, 2006, *312* (5782), 1900–1902.
- -, "The economics, technology, and neuroscience of human capability formation," Proceedings of the national Academy of Sciences, 2007, 104 (33), 13250–13255.
- Imdad, Aamer, Mohammad Yawar Yakoob, Christopher Sudfeld, Batool A Haider, Robert E Black, and Zulfiqar A Bhutta, "Impact of vitamin A supplementation on infant and childhood mortality," BMC Public Health, 2011, 11 (Suppl 3), S20.
- Klemm, R D W, A B Labrique, P Christian, M Rashid, A A Shamim, J Katz, A Sommer, and K P West, "Newborn Vitamin A Supplementation Reduced Infant Mortality in Rural Bangladesh," *PEDI-ATRICS*, July 2008, *122* (1), e242–e250.

- Kling, Jeffrey R, Jeffrey B Liebman, and Lawrence F Katz, "Experimental analysis of neighborhood effects," *Econometrica*, 2007, 75 (1), 83–119.
- Kuczmarski, Robert J, Cynthia L Ogden, Laurence M Grummer-Strawn, Katherine M Flegal, Shumei S Guo, Rong Wei, Zuguo Mei, Lester R Curtin, Alex F Roche, and Clifford L Johnson, "CDC growth charts: United States.," *Advance data*, 2000, (314), 1.
- Labrique, Alain B, Parul Christian, Rolf DW Klemm, Mahbubur Rashid, Abu A Shamim, Allan Massie, Kerry Schulze, Andre Hackman, and Keith P West, "A cluster-randomized, placebo-controlled, maternal vitamin A or beta-carotene supplementation trial in Bangladesh: design and methods," *Trials*, 2011, *12* (1), 102.
- Maccini, Sharon and Dean Yang, "Under the Weather: Health, Schooling, and Economic Consequences of Early-Life Rainfall," *The American Economic Review*, June 2009, *99* (3), 1006–1026.
- Persson, Petra and Maya Rossin-Slater, "Family Ruptures and Intergenerational Transmission of Stress," Technical Report 2014.
- Stephensen, Charles B, "Vitamin A, infection, and immune function*," *Annual review of nutrition*, 2001, *21* (1), 167–192.
- Sugimoto, Jonathan D, Alain B Labrique, Salahuddin Ahmad, Mahbubur Rashid, Abu Ahmed Shamim, Barkat Ullah, Rolf D W Klemm, Parul Christian, and Keith P West, "Epidemiology of tornado destruction in rural northern Bangladesh: risk factors for death and injury.," *Disasters*, April 2011, 35 (2), 329–345.
- Watson, Tara, "Public health investments and the infant mortality gap: Evidence from federal sanitation interventions on US Indian reservations," *Journal of Public Economics*, 2006, *90* (8), 1537–1560.
- West, Keith P, Parul Christian, Alain B Labrique, Mahbubur Rashid, Abu Ahmed Shamim, Rolf DW Klemm, Allan B Massie, Sucheta Mehra, Kerry J Schulze, and Hasmot Ali, "Effects of vitamin A or beta carotene supplementation on pregnancy-related mortality and infant mortality in rural Bangladesh," JAMA, 2011, 305 (19), 1986.
- Zeba, Augustin N, Hermann Sorgho, Noël Rouamba, Issiaka Zongo, Jeremie Rouamba, Robert T Guiguemdé, Davidson H Hamer, Najat Mokhtar, and Jean-Bosco Ouedraogo, "Major reduction of

malaria morbidity with combined vitamin A and zinc supplementation in young children in Burkina Faso: a randomized double blind trial," *Nutr J*, 2008, 7 (7), 7.

								I		M	ithin Torr	nado Are	я	
	IA I I = N	l 269	Torn N =	ado 345	Non-to N = 4	rnado 1924	Diffe	rence	Vitam	in A 184	Place N = 1	bo 161	Differ	ence
	Mean	SD	Mean	SD	Mean	SD	Mean	SE	Mean	SD	Mean	SD	Mean	SE
Infant birth anthropometry														
Weight (kg)	2.49	0.43	2.51	0.41	2.49	0.43	0.02	0.03	2.54	0.38	2.49	0.43	0.04	0.06
Height (cm)	46.67	2.35	46.61	2.23	46.68	2.36	-0.06	0.17	46.80	2.09	46.40	2.37	0.40	0.31
MUAC (cm)	9.44	0.85	9.48	0.81	9.43	0.85	0.04	0.06	9.50	0.73	9.44	0.89	0.06	0.11
Head Circumference (cm)	32.67	1.62	32.73	1.71	32.67	1.61	0.06	0.12	32.82	1.58	32.63	1.84	0.19	0.24
Chest Circumference (cm)	30.74	2.10	30.77	1.96	30.73	2.11	0.03	0.15	30.80	1.74	30.73	2.18	0.07	0.27
Infant anthropometry at 3 months														
MUAC (cm)	12.38	1.06	12.43	1.07	12.37	1.06	0.05	0.06	12.43	1.02	12.42	1.14	0.01	0.12
Head Circ. (cm)	38.71	1.48	38.58	1.45	38.72	1.48	-0.14	0.09	38.51	1.33	38.67	1.59	-0.16	0.17
Chest Circumference (cm)	38.89	2.21	38.98	2.17	38.89	2.22	0.10	0.13	38.82	2.15	39.18	2.18	-0.36	0.25
Anthropometric Index	0.17	0.98	0.17	0.97	0.17	0.99	0.00	0.06	0.13	0.91	0.22	1.05	-0.09	0.11
Infant anthropometry at 6 months														
MUAC	13.09	1.04	13.20	1.06	13.09	1.04	0.11	0.06*	13.27	1.05	13.12	1.07	0.15	0.12
Head Circumference (cm)	40.89	1.41	41.01	1.38	40.88	1.42	0.13	0.08	40.97	1.41	41.05	1.36	-0.08	0.16
Chest Circumference (cm)	41.34	2.11	41.59	2.11	41.32	2.11	0.27	0.12^{**}	41.51	2.13	41.67	2.10	-0.16	0.24
Anthropometric Index	0.04	0.99	0.16	1.00	0.03	0.98	0.13	0.06^{**}	0.17	1.02	0.16	0.99	0.01	0.11
Other infant outcomes														
Gender is Male	0.51	0.50	0.54	0.50	0.51	0.50	0.03	0.03	0.52	0.50	0.57	0.50	-0.06	0.05
Fever Incidence, 0-3 months	0.58	0.49	0.56	0.50	0.59	0.49	-0.03	0.03	0.56	0.50	0.55	0.50	0.01	0.06
Fever Incidence 0-6 months	0.55	0.50	0.57	0.50	0.55	0.50	0.02	0.03	0.57	0.50	0.57	0.50	0.00	0.06
Mortality 0-24 weeks	0.05	0.22	0.04	0.20	0.05	0.23	-0.01	0.01	0.04	0.21	0.04	0.19	0.01	0.02
Maternal characteristics														
Parity	1.33	2.41	1.23	1.48	1.33	2.46	-0.10	0.13	1.38	1.64	1.07	1.26	0.31	0.16^{*}
ISI	-0.04	0.99	-0.11	0.94	-0.04	0.99	-0.07	0.06	-0.09	0.97	-0.13	0.90	0.04	0.10
Height (cm)	149.32	5.15	149.01	5.08	149.34	5.16	-0.34	0.29	148.72	4.97	149.33	5.20	-0.61	0.55
MUAC (cm)	22.66	1.93	22.61	2.00	22.66	1.93	-0.05	0.11	22.58	2.08	22.65	1.91	-0.07	0.22
Education (years)	3.62	4.03	3.54	3.94	3.62	4.04	-0.08	0.22	3.92	4.26	3.11	3.50	0.81	0.42*
Dosing														
$Dosed \le 6 hours$	0.49	0.50	0.55	0.50	0.49	0.50	0.07	0.03^{**}	0.50	0.50	0.62	0.49	-0.12	0.06*
Dosed <= 12 hours	0.63	0.48	0.69	0.46	0.63	0.48	0.06	0.03*	0.65	0.48	0.73	0.45	-0.07	0.06
$Dosed \le 18 hours$	0.71	0.45	0.75	0.43	0.71	0.46	0.04	0.03	0.71	0.45	0.80	0.40	-0.08	0.06
$Dosed \le 24 hours$	0.75	0.43	0.78	0.42	0.75	0.43	0.03	0.03	0.75	0.43	0.81	0.39	-0.06	0.05
$Dosed \le 7 days$	0.85	0.35	0.88	0.32	0.85	0.36	0.03	0.02	0.88	0.32	0.88	0.32	0.00	0.04
Summary statistics for the study sample (infant san	iple), limi	ted to infar	its born a	t least 9 mc	onths befo	re the torn	ado. Torna	do and Noi	n-Tornado	refer to in	a of the v	outside the	tornado the data
rather than as formal hypothesis tests (whi	ich would	need to ta	uke account	of cluste	ring).		10110 nm		ern p-vaures	IN STACE				uic data
Significance: * < 0.1; ** < 0.05; *** < 0.0	11.				Ì									

Table 1: Summary Statistics of Infants in the Pre-tornado Cohorts

27

	-
	9
	0
	d,
	Η.
	0
	-
•	·
	ē.
,	9
	-
•	
	5
	Ξ.
	ຼ
	Ξ.
	Ξ.
Ť	-
1	
	دە
	Ē
	₽
	ਛ
	-
	9
	Η.
	۲
H	5
÷	<u> </u>
	7
	F
	s
	Ē
	F
	d
•	Ξ.
	육
	-
	-
	ö
•	÷.
	ŝ
	÷.
	Ξ.
	• •
	6
	ta
c	Sta
c	t Sta
¢	nt Sta
c	ent Sta
c	nent Sta
c	ment Sta
c	utment Sta
C	atment Sta
C	eatment Sta
•	reatment Sta
ſ	I reatment Sta
E	1 I reatment Sta
Ę	nd I reatment Sta
Ę	ind I reatment Sta
E	and Ireatment Sta
E	o and I reatment Sta
E	to and I reatment Sta
E	ado and Ireatment Sta
E	nado and Ireatment Sta
E F	inado and Ireatment Sta
Ę	ornado and Ireatment Sta
E F	cruado and Ireatment Sta
E	Iornado and Ireatment Sta
E	y lornado and Ireatment Sta
E	by Iornado and Ireatment Sta
E E	by Iornado and Ireatment Sta
E E	g by Iornado and Ireatment Sta
E E	ng by Iornado and Ireatment Sta
E E	ang by Iornado and Ireatment Sta
E F F	osing by Iornado and Ireatment Sta
E E	osing by Iornado and Ireatment Sta
E E E	Uosing by Iornado and Ireatment Sta
E E E	Dosing by Iornado and Ireatment Sta
E E	o Dosing by Iornado and Ireatment Sta
E E E	to Dosing by Iornado and Ireatment Sta
	e to Dosing by Iornado and Ireatment Sta
H H H	ne to Dosing by Iornado and Ireatment Sta
H	me to Dosing by Iornado and Ireatment Sta
	ime to Dosing by Iornado and Ireatment Sta
H H H	I ime to Dosing by Iornado and I reatment Sta
e H H H	: 1 ime to Dosing by Iornado and Ireatment Sta
· · · · · · · · · · · · · · · · · · ·	2: I ime to Dosing by Iornado and Ireatment Sta
· · · · · · · · · · · · · · · · · · ·	2: Lime to Dosing by Lornado and Ireatment Sta
H H H	le 2: 1 ime to Dosing by Iornado and Ireatment Sta
H H H	ble 2: 1 ime to Dosing by Iornado and Ireatment Sta
H · · · · · · · · · · · · · · · · · · ·	able 2: 1 ime to Dosing by Iornado and Ireatment Sta
H · · · · · · · · · · · · · · · · · · ·	lable 2: 1 ime to Dosing by Iornado and I reatment Sta

Within Tornado Area

SE		0.05	0.05	0.05	0.05	0.04	-
Mean		-0.03	-0.02	-0.04	-0.04	-0.05	
SD		0.50	0.48	0.45	0.42	0.33	-
Mean		0.50	0.65	0.73	0.77	0.87	
SD		0.50	0.49	0.46	0.44	0.38	E
Mean		0.47	0.62	0.69	0.74	0.82	-
SE		0.03	0.03	0.03	0.02	0.02	1. 1.
Mean		0.01	0.00	0.00	0.01	0.00	-
SD		0.50	0.48	0.45	0.43	0.36	۰ ۱
Mean		0.47	0.64	0.71	0.75	0.84	-
SD		0.50	0.48	0.45	0.43	0.36	
Mean		0.48	0.63	0.71	0.75	0.85	F
SD		0.50	0.48	0.45	0.43	0.36	-
Mean		0.47	0.64	0.71	0.75	0.84	-
	Dosing	Dosed $\leq = 6$ hours	Dosed <= 12 hours	Dosed <= 18 hours	Dosed $\leq = 24$ hours	$Dosed \le 7 days$	
	Mean SD Mean SD Mean SD Mean SE Mean SD Mean SD Mean SE	Mean SD Mean SD Mean SD Mean SD Mean SD Mean SE Mean <td>$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$</td> <td></td>	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	

Summary statistics of time-to-dosing by area. Tornado and Non-tornado refer to inside vs. outside the tornado area. The last three columns restrict the sample to only within the tornado area. The sample includes only infants that were in-utero when the tornado hit (whether inside or outside the affected area). Significance: * < 0.1, ** < 0.05; *** < 0.01.

	E.		- -			
	Dirun m Missing	easures Missing	o monun Missine	measures Missing	o monun 1 Missing	measures Missing
	0	or late	Missing	or late	0	or late
	b/se	b/se	b/se	b/se	b/se	b/se
Triple interaction: In tornado area X Vit A X		0		0	0	
0-270 days after LMP	-0.00	0.04	-0.00	-0.01	-0.04	-0.05 10.05
271-360 days after LMP	(10.0)	(00.0)	0.00	00.0	(0.01)	0.00
`			(0.06)	(0.06)	(0.06)	(0.07)
361-450 days after LMP					-0.01	-0.04
Double interaction: In tornado area X					(0.06)	(0.06)
0-270 days after LMP	0.04	0.00	0.03	0.03	0.07**	0.07**
271-360 days after LMP	(0.03)	(0.04)	(0.03) -0.01	(0.03) -0.02	(0.03) -0.00	(0.03) 0.00
			(0.04)	(0.04)	(0.04)	(0.05)
301-430 days after LMP					-0.01	-0.00
Double interaction: Vit A X					(20.0)	(00.0)
0-270 days after LMP	-0.00	0.00	0.01	0.01	0.00	0.00
271-360 days after LMP	(10.0)	(10.0)	(0.01) 0.03*	(0.01) 0.04^{**}	(0.01) 0.02	(0.01) 0.02
			(0.02)	(0.02)	(0.02)	(0.02)
361-450 days after LMP					0.00	-0.00
Cohorts:					(70.02)	(70.02)
0-270 days after LMP	-0.12***	-0.10^{***}	-0.03***	-0.03***	-0.02***	-0.03***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
271-360 days after LMP			-0.03**	-0.03***	-0.02*	-0.02*
361-450 dave a fret 1 MD			(10.0)	(10.0)	(10.0) -0.00	(10.0) -0.00
201 120 mays area area					(0.01)	(0.01)
Observations	18872	18872	18872	18872	18872	18872
Attrition in the data by cohort. The dependent variable in column. 3-month anthropometry and 6 month-anthropometry. The depend that infants measured late (after 7 days for birth anthropometry, also coded as 1. Our main outcome measures used in the paper a correspond to the attrition for those main outcome measures. By the hierblate (which is mission for monot f, the inform the streic)	1, 3 and 5 is ent variable i or 8 weeks a ce set to miss cause or nor	a dummy in n columns 2 fiter the targ ing after the mal cohort	dicating mis , 4, and 6 is fet date for ese cutoff di definitions (sing values f the same as 3 and 6 moi ates so the e (used throug	or birth anth the odd colu ath anthrope ven number hout the pa	rropometry, mns except ometry) are ed columns per) rely on
since the last menstrual period to fit approximately the in-utero (0- periods. Standard errors are clustered at the sector (treatment rand Significance: $* < 0.10$, $** < 0.05$, $*** < 0.01$.	.70 days), firs omization) le	t three mont vel.	hs (271-360	days) and 3-	6 month (36	1-450 days)

Table 3: Attrition at birth, 3 months and 6 months

		Share	of houses de	estroved		
	0-20%	21-40%	41-60%	61-80%	81-100%	Total
Control (# of sectors):	6	4	, -	6	9	20
Vitamin A (# of sectors):	11	1	7	2	0	21
All tornado damaged sectors						
Average Control:	47.6%					
Average Vitamin A:	33.7%					
Vitamin A - Control	= -14%					
Excluding sectors with > 80°	% damage					
Average Control:	28.7%					
Average Vitamin A:	33.5%					
Vitamin A - Control :	= 5%					
Tornado damage (share of houses	destroyed) by	treatment all	ocation (with	in sectors wit	h any damage	

Table 4: Contingency Table of Tornado Damage by Treatment Status

	BW	log(BW)	LBW	BGGA	ΡM	SMGA	Height	MUAC	HC	CC	BirthAI
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se
In tornado area X In utero	-0.13**	-0.05**	0.15*	-0.38	0.04	0.03	-0.55**	-0.25**	-0.47*	-0.29	-0.27**
	(0.06)	(0.02)	(0.08)	(0.35)	(0.04)	(0.04)	(0.23)	(0.12)	(0.27)	(0.19)	(0.13)
In tornado area X In utero X Maternal supplementation	0.03	0.01	-0.04	0.52	-0.06	0.06	-0.06	0.06	0.10	0.07	0.04
	(0.07)	(0.03)	(0.0)	(0.44)	(0.05)	(0.00)	(0.34)	(0.15)	(0.34)	(0.25)	(0.16)
In utero X Maternal supplementation	0.01	0.01	0.00	-0.13	0.04^{**}	-0.02	-0.05	0.01	-0.01	0.01	0.01
	(0.02)	(0.01)	(0.02)	(0.12)	(0.02)	(0.02)	(0.12)	(0.04)	(0.10)	(0.07)	(0.05)
In utero	0.01	0.00	-0.01	0.45***	-0.07***	-0.00	0.32^{***}	-0.05	-0.01	0.01	0.02
	(0.02)	(0.01)	(0.02)	(0.0)	(0.02)	(0.01)	(0.10)	(0.03)	(0.08)	(0.06)	(0.04)
Age at newborn anthropometry	-0.01 **	-0.00***	0.01*	-0.00	0.00	-0.00***	0.05 **	-0.08***	-0.06***	0.08^{***}	-0.02**
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.02)	(0.01)	(0.02)	(0.01)	(0.01)
Constant	2.44***	0.88^{***}	0.54^{***}	37.61^{***}	0.29^{***}	0.26^{***}	46.36^{***}	9.38***	30.46^{***}	32.32***	0.03^{***}
	(0.00)	(0.00)	(00.0)	(0.02)	(0.00)	(0.00)	(0.02)	(0.01)	(0.02)	(0.02)	(0.01)
Observations	13884	13884	13884	16750	16750	16259	13521	13788	13660	13769	13897
Linear regression models of birth outcomes as a function of ton β -carotene supplementation groups as they are similar and had si is birth weight in kilograms. log(BW) is the log of birth weight. I age < 37 weeks). Small for Gestational Age (SMGA) is an indicate circumference (MUAC), head circumference (HC) and chest circumference values of: birth weight, birth height, birth MUAC, birth HC and bi Significance: $* < 0.10, ** < 0.05, *** < 0.01$.	rnado exposi milar effects JBW is an in or that is equ inference are	are in utero, $(\beta$ -carotene dicator for l al to 1 if the measured i dard errors	interacted can be cor birth weigh infant is b n centimet are clustere	with vitamir werted into v t < 2.5kg. G elow the 25th ers. BirthAI at the secto	a A or β -ca vitamin A b A is our me n percentile is an index or level.	rotene supp y the body)) asure of ge in weight gi constructed	lementation). The samp stational age ven gestatio as the stan	of the motl ole includes c . PM is an ii nal age (for t dardized (0 n	her (we com only live bor: ndicator for this sample). nean, 1 SD)	bine the vit: a singleton i prematurity Height, mic average of s	umin A and nfants. BW (gestational -upper arm tandardized

outcomes
birth
u0
supplementation
maternal
and
exposure
tornado
\mathbf{of}
e effect
$\mathbf{T}_{\mathbf{h}}$
Table

	- - -	1	
41	Not in Lornado V	Vindow	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	-0.00	-0.01	-0.01 (0.02)
In Tomado Area	0.14	0.09	-0.06 (0.08)
Difference	$0.14(0.05)^{***}$	0.10 (0.06)*	-0.05 (0.08)
	In utero at tori	<u>nado</u>	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	-0.01	-0.00	0.01 (0.04)
In Tornado Area	-0.01	0.09	0.10(0.15)
Difference	-0.01 (0.09)	0.09 (0.12)	0.10(0.15)
	first 10 weeks at	: tornado	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	0.01	-0.02	-0.03 (0.05)
In Tornado Area	-0.10	0.13	0.23 (0.16)
Difference	-0.11 (0.14)	$0.15(0.08)^{*}$	0.26(0.16)
Difference in-utero	-0.15 (0.10)	-0.01 (0.13)	0.14 (0.16)
Difference in first 10 weeks	-0.26 (0.14)*	$0.05\ (0.10)$	$0.31 (0.18)^{*}$
The table shows means and diff.	erences (along wit)	h standard erroi	s of differences) for the
various subsamples that define th	he triple-difference	strategy: within	i vs. outside the tornado
at the time of tornado) and in t	reatment versus co	ontrol sectors.	The outcome variable is
the anthropometric index (stand	ardized average of	f MUAC, head	circumference and chest
circumference) measured at 3 mor	nths of age (the ind	lex SD is 1). Star	adard errors are clustered
at the sector (vitamin A randomi	zation) level. Signi	ficance: $* < 0.1$	** < 0.05; *** < 0.01.

Table 6: Difference-in-Difference Table: Anthropometric index at 3 months.

	Not in Tornado	Window	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	0.01	-0.00	-0.01 (0.03)
In Tornado Area	0.18	0.14	-0.04 (0.07)
Difference	0.17 (0.06)***	0.15 (0.05)***	-0.03 (0.08)
	In utero at to	<u>rnado</u>	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	-0.02	-0.02	-0.01 (0.04)
In Tornado Area	0.08	0.17	0.09 (0.12)
Difference	*(90.0) 60.0	$0.19 (0.11)^{*}$	0.09 (0.12)
	n first 22 weeks :	at tornado	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	-0.02	-0.02	-0.01 (0.04)
In Tornado Area	-0.06	0.17	0.23 (0.12)*
Difference	-0.04(0.10)	0.20 (0.08)**	$0.24 \ (0.13)*$
Difference in-utero	-0.08 (0.07)	0.04 (0.10)	0.12 (0.12)
Difference in first 22 weeks	-0.21 (0.10)**	0.05 (0.09)	$0.26 (0.14)^{*}$
The table shows means and diffe	rences (along with	standard errors of	f differences) for the var-
ious subsamples that define the t within versus outside the tornado	riple-difference str window (i.e. in ut	rategy: within vs. eeto or in first 22 w	outside the tornado area; eeks of life at the time of
tornado) and in treatment versus	control sectors. T	'he outcome varial	ole is the anthropometric
index (standardized average of M	UAC, head circum	ference and chest	circumference) measured
at 6 months of age (the index SL) is 1). Standard e $e^{-k} < 0.1 \cdot ** < 0.1$	rrors are clustered 15: *** < 0.01	at the sector (vitamin A
TAILUULINEALUULI ILVUL JIENUL	C	17°	

Table 7: Difference-in-Difference Table: Anthropometric index at 6 months.

	A		IW	IAC.				
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se
Triple interaction: In tornado area X Vit A X								
In utero	0.20	0.17	0.22	0.20	0.43	0.40	0.20	0.17
	(0.15)	(0.15)	(0.15)	(0.15)	(0.38)	(0.36)	(0.23)	(0.22)
0-10 weeks of age	0.41 **	0.32*	0.44 **	0.34*	0.87 **	0.74*	0.38	0.30
)	(0.19)	(0.19)	(0.19)	(0.19)	(0.43)	(0.43)	(0.28)	(0.27)
Double interaction: In tornado area X								
In utero	-0.17	-0.12	-0.19*	-0.14	-0.52*	-0.40	-0.07	-0.01
	(0.11)	(0.10)	(0.10)	(0.09)	(0.27)	(0.25)	(0.18)	(0.17)
0-10 weeks of age	-0.31 **	-0.30**	-0.33**	-0.30 **	-0.87**	-0.85**	-0.16	-0.15
)	(0.15)	(0.15)	(0.15)	(0.15)	(0.36)	(0.37)	(0.22)	(0.20)
Double interaction: Vit A X								
In utero	0.01	0.02	-0.01	-0.01	0.05	0.06	0.05	0.06
	(0.04)	(0.04)	(0.04)	(0.04)	(0.0)	(0.08)	(0.05)	(0.05)
0-10 weeks of age	-0.02	-0.02	-0.05	-0.04	-0.06	-0.07	0.05	0.06
)	(0.05)	(0.05)	(0.06)	(0.06)	(0.13)	(0.12)	(0.08)	(0.08)
Cohorts:								
In utero	-0.03	-0.05**	-0.03	-0.06**	0.01	-0.04	-0.08**	-0.11^{***}
	(0.03)	(0.03)	(0.03)	(0.03)	(0.00)	(0.06)	(0.04)	(0.04)
0-10 weeks of age	0.05	0.04	0.06	0.05	0.05	0.04	0.04	0.03
1	(0.04)	(0.03)	(0.04)	(0.04)	(0.0)	(0.08)	(0.06)	(0.06)
- Other controls -								
Best guess length of gestation (weeks)	0.06***	0.05***	0.05***	0.04^{***}	0.13^{***}	0.11^{***}	0.09^{***}	0.08^{***}
	(0.00)	(0.00)	(0.00)	(0.00)	(0.01)	(0.01)	(0.00)	(0.00)
Age at 3 month measurement (centered)	0.02^{***}	0.02^{***}	0.02^{***}	0.02^{***}	0.04^{***}	0.04^{***}	0.04^{***}	0.04^{***}
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Maternal MUAC		0.06***		0.07^{***}		0.13^{***}		0.06^{***}
		(0.00)		(0.00)		(0.01)		(0.01)
Maternal height		0.03***		0.03^{***}		0.07^{***}		0.04^{***}
		(0.00)		(0.00)		(0.00)		(0.00)
Living Standards Index		0.10^{***}		0.10^{***}		0.21^{***}		0.11^{***}
		(0.01)		(0.01)		(0.02)		(0.01)
Observations	16490	16467	16471	16448	15970	15950	16449	16426
Linear regression models of infant development measured by ant head circumference (HC) and chest circumference (CC). all measu	hropometry at ured in centim	3 months.	. The outco	ome variable ometric inde	es are mid-u ex (AI) that	upper arm ci is a standar	ircumfereno dized (zero	ce (MUAC), mean, unit
SD) average of the three variables after each has been standardize	ed to zero mea	in and unit	standard d	leviation. "	Vit A'' is an	indicator th	nat is 1 if in	fants in the

Table 8: Anthropometry at 3 months: Main specification (triple-difference).

contains randomization sector fixed effects (this absorbs main effects of the tornado area and treatment indicators). Living Standards is an index based on a principal components analysis of household assets. Best guess gestational length is based on date of last menstrual period reported at enrollment in the study sector were given vitamin A and zero if they were in the placebo group. "In Tornado Area" is an indicator defined as 1 if any households in the sector were destroyed in the tornado. "In utero" is an indicator that is 1 if the infant was in utero at the date of the tornado and "0-10 weeks of age" is 1 if the infant was in his or her first 10 weeks of life at the date of the tornado. Combined these three types of indicators define our triple difference strategy. Each regression (in the first trimester). Standard errors are clustered at the sector (treatment randomization) level. Significance: * < 0.10; ** < 0.05; *** < 0.01. SD head Line

Table 9: Anthropometry at 6 months: Main specification (triple-difference).

	A	Ι	MC	IAC	Õ	C	Ĥ	0
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se
Triple interaction: In tornado area X Vit A X								
In utero	0.17	0.14	0.28*	0.26*	0.25	0.20	0.08	0.04
	(0.13)	(0.12)	(0.14)	(0.14)	(0.29)	(0.27)	(0.20)	(0.20)
0-10 weeks of age	0.52^{***}	0.42^{**}	0.67***	0.56^{***}	0.71*	0.54	0.57**	0.47*
0	(0.18)	(0.18)	(0.17)	(0.17)	(0.42)	(0.39)	(0.27)	(0.26)
11-22 weeks of age	0.12	0.13	0.28	0.29	0.19	0.22	-0.06	-0.06
	(0.19)	(0.20)	(0.20)	(0.22)	(0.38)	(0.40)	(0.27)	(0.28)
Double interaction: In tornado area X								
In utero	-0.10	-0.04	-0.11	-0.07	-0.20	-0.10	-0.07	-0.01
	(0.08)	(0.07)	(0.09)	(0.09)	(0.18)	(0.16)	(0.13)	(0.13)
0-10 weeks of age	-0.34**	-0.31**	-0.40***	-0.37***	-0.62*	-0.58*	-0.26	-0.24
	(0.14)	(0.13)	(0.12)	(0.11)	(0.33)	(0.30)	(0.22)	(0.20)
11-22 weeks of age	-0.15	-0.16	-0.26*	-0.28*	-0.38	-0.41*	0.09	0.09
;	(0.13)	(0.13)	(0.15)	(0.15)	(0.24)	(0.24)	(0.19)	(0.19)
Double interaction: Vit A X	0	0	0	•		0	0	
In utero	0.00	0.01	-0.04	-0.03	0.02	0.04	0.04	c 0.0
	(0.04)	(0.04)	(0.04)	(0.04) 0.14	(0.09)	(0.09)	(0.06)	(0.06)
U-1U WEEKS OF AGE	-0.04	CU.U-	-0.11*	-0.11*	-0.10	-0.11	/0.0	80.0
11 22 modes of area	(00.0)	(cn.u)	(00.0)	(00.0)	(0.1.0) 0.03	(0.12) 0.04	(60.0)	(60.0)
11-22 weeks of age	-0.00	0.00	-0.02	-0.02	-0.05 (0.10)	-0.04	c0.0	00
Cohorts:	(c0.0)	(c0.0)	(c0.0)	(cn.u)	(0.12)	(0.11)	(/0.0)	(0.07)
In intern	-0.03	-0.07**	-0 06*	***6U U-	0.04	-0.03	-0.04	-0.08*
	(0.03)	(0.03)	(0.03)	(0.03)	(0.06)	(90.0)	(0.04)	(0.04)
0-10 weeks of age	0.05	0.04	0.04	0.03	0.13	0.11	0.02	0.01
c	(0.04)	(0.04)	(0.04)	(0.04)	(0.0)	(0.08)	(0.00)	(0.00)
11-22 weeks of age	-0.06	-0.07*	-0.10^{***}	-0.11***	-0.14*	-0.16^{**}	-0.01	-0.01
0	(0.04)	(0.04)	(0.04)	(0.04)	(0.08)	(0.08)	(0.05)	(0.05)
- Other controls -	~	~	~	~	~	~	~	~
Best guess length of gestation (weeks)	0.04^{***}	0.03^{***}	0.03^{***}	0.02^{***}	0.08^{***}	0.06***	0.05***	0.04^{***}
	(0.00)	(0.00)	(0.00)	(0.00)	(0.01)	(0.01)	(0.00)	(0.00)
Age at 6 month measurement (centered)	0.01***	0.01***	0.00***	0.00***	0.02^{***}	0.02^{***}	0.02***	0.02***
Maternal MITAC	(0.00)	(0.00) 0.07***	(0.00)	(0.00) 0.08***	(0.00)	(0.00) 0.15***	(0.00)	(0.00) 0.06***
		(00.00)		0000		(0.01)		0.01)
Maternal height		0.03***		0.02***		0.07***		0.04***
)		(0.00)		(0.00)		(0.00)		(0.00)
Living Standards Index		0.12^{***}		0.12^{***}		0.25^{***}		0.11^{***}
		(0.01)		(0.01)		(0.02)		(0.01)
Observations	16226	16202	16211	16187	15578	15555	16118	16094
Linear regression models of infant development measured by a	nthropometry a	t 6 months	S. The outco	me variables	are mid-up	per arm ci	rcumferenc	e (MUAC),
head circumference (HC) and chest circumference (CC), all mer	asured in centin	neters and	an anthropo	metric index	(AI) that i	s a standar	dized (zero	mean, unit
D) average of the three variables after each has been standard	Ized to zero me	an and uni	t standard d	eviation. V	it A' is an i - 1 1 :6 -	ndicator th	at is 1 st ini 	ants in the
sector were given vitanin A and zero it mey were in me piaced destroyed in the tornado. "In utero" is an indicator that is 1 if f	oo group. In I he infant was in	ornauo Ar utero at th	ea is an inc ne date of th	ncator denna le tornado ai	ed as 1 11 a nd "0-10 we	ny nousenc seks of age	its in the s is 1 if the	ector were infant was

uccurrent in this or her first 10 weeks of life at the date of the tornado. Combined these three types of indicators define our triple difference strategy. Each regression contains randomization sector fixed effects (this absorbs main effects of the tornado area and treatment indicators). Living Standards is an index based on a principal components analysis of household assets. Best guess gestational length is based on date of list trimester). Standards is an index based on a principal components analysis of household assets. Best guess gestational length is based on date of last menstrual period reported at enrollment in the study (in the first trimester). Standard errors are clustered at the sector (treatment randomization) level. Significance: * < 0.10; ** < 0.05; *** < 0.01.

	Not in Tornado V	Window	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	0.92	0.91	-0.01 (0.02)
In Tornado Area	0.89	0.92	0.03 (0.09)
Difference	-0.03 (0.07)	0.01 (0.05)	0.04(0.09)
	In utero at tor	<u>nado</u>	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	0.87	0.87	-0.01 (0.04)
In Tornado Area	1.10	1.21	0.11 (0.18)
Difference	$0.22 (0.11)^{**}$	0.34 (0.15)**	0.12(0.18)
II	ı first 10 weeks a	t tornado	
	Placebo	Vitamin A	Difference (VA-PL)
	b (se)	b (se)	b (se)
Not in Tornado Area	0.86	0.88	0.02 (0.06)
In Tornado Area	1.34	0.88	-0.46 (0.25)*
Difference	$0.48(0.19)^{**}$	-0.01 (0.16)	-0.48 (0.25)*
Difference in-utero	$0.26\ (0.11)^{**}$	0.34 (0.14)**	0.08 (0.18)
Difference in first 10 weeks	0.51 (0.16)***	-0.01 (0.16)	-0.52 (0.23)**
The table shows means and diff various subsamples that define th area; within versus outside the to time of tornado) and in treatmen	ferences (along wi he triple-difference rnado window (i.e. t versus control se	th standard error e strategy: within . in utero or in fu ctors. The outco	s of differences) for the t vs. outside the tornado est 10 weeks of life at the me variable is number of
rever episodes in the first 2 mont are clustered at the sector (vitam	ns top coaea at 4 (in A randomizatio	✓ 4 episodes are n) level. Signific	set to 4). Standard errors ance: $* < 0.1$; $** < 0.05$;
*** < 0.01.			

Table 10: Difference-in-Difference Table: Fever episodes at 0-3 months.

	In m	onths 0-3	In me	onths 4-6
	Any fever b/se	Num. episodes b/se	Any fever b/se	Num. episodes b/se
Triple interaction: In tornado area X Vit A X				
In utero	0.00	0.05	-0.03	0.03
	(0.07)	(0.18)	(0.08)	(0.17)
0-3 months of age	-0.14	-0.39**	-0.05	-0.18
)	(0.10)	(0.19)	(0.10)	(0.27)
4-6 months of age			-0.01	-0.11
-			(0.00)	(0.24)
Double interaction: In tornado area X In interio	0.07	***℃ ∪	0.05	0.16
	(0.05)	(0.11)	(0.05)	(0.12)
0-3 months of age	0.22^{***}	0.53 * * *	0.10	0.44 **
0	(0.07)	(0.12)	(0.07)	(0.21)
4-6 months of age			0.05	0.42^{**}
			(0.08)	(0.17)
Double interaction: Vit A X				
In utero	-0.03	0.02	0.00	-0.02
	(0.02)	(0.04)	(0.02)	(0.04)
0-3 months of age	-0.04	0.00	0.03	0.00
	(0.03)	(0.05)	(0.03)	(0.06)
4-6 months of age			0.01	-0.02
			(0.03)	(0.05)
Lo ritorio	000	**700		
TII ULETO	-0.00		20.0	0.04
0-3 months of age	(10.0) -0.01	(cn·n) *90.0-	(10.01) 0.03	(cn·n) **60 ⁻ 0
. 0.	(0.02)	(0.03)	(0.02)	(0.04)
4-6 months of age		~	0.01	0.05
) .			(0.02)	(0.04)
- Other controls - Best guess length of gestation (weeks)	0.00	-0.00	-0.00	-0.00
0	(000)	(000)	(0.00)	(0.00)
Age at 3 month measurement (centered)	0.00 ***	0.01 ***	~	~
	(0.00)	(0.00)		
Age at 6 month measurement (centered)			-0.00	-0.00
			(0.00)	(0.00)

Table 11: Incidence of fever: main specification (triple-difference)

Linear regression models of fever using our main specifications. The outcome variables of fever episodes in 0-3 months and 4-6 months are top coded at 4 (>4 episodes are coded as 4). Significance: * < 0.10; ** < 0.05, *** < 0.01.

Observations

post-tornado cohort.
g
defining
5
regressions
key
f
Robustness (
άi
Table 12

	Α	_	M	JAC	Fev	'er
	3 Mo.	6 Mo.	3 Mo.	6 Mo.	3 Mo.	6 Mo.
	b/se	b/se	b/se	b/se	b/se	b/se
Triple int.: In tornado area X Vit A X						
Conceived after tornado	0.04	-0.13	-0.02	-0.21	-0.01	0.04
	(0.13)	(0.15)	(0.15)	(0.17)	(0.17)	(0.20)
In utero	0.20	0.11	0.21	0.19	0.04	0.03
	(0.16)	(0.15)	(0.17)	(0.17)	(0.19)	(0.17)
0-10 weeks of age	0.42 **	0.47^{**}	0.43^{**}	0.58^{***}	-0.47*	-0.21
	(0.20)	(0.19)	(0.20)	(0.18)	(0.24)	(0.27)
11-22 weeks of age		0.06		0.19		-0.16
•		(0.20)		(0.22)		(0.26)
Double interaction: In tornado area X						
Conceived after tornado	0.23^{***}	0.19*	0.22^{***}	0.14	-0.04	-0.13
	(0.08)	(0.11)	(0.0)	(0.13)	(0.11)	(0.15)
In utero	-0.09	-0.01	-0.11	-0.05	0.29^{***}	0.11
	(0.11)	(0.10)	(0.11)	(0.12)	(0.11)	(0.14)
0-10 weeks of age	-0.24	-0.26*	-0.25	-0.34***	0.45**	0.37*
	(0.16)	(0.15)	(0.16)	(0.12)	(0.18)	(0.22)
11-22 weeks of age		-0.07		-0.20		0.49^{***}
		(0.15)		(0.17)		(0.18)
Double interaction: Vit A X						
Conceived after tornado	-0.04	-0.06	-0.03	-0.06	0.02	0.02
	(0.04)	(0.05)	(0.05)	(0.05)	(0.05)	(0.05)
In utero	-0.00	-0.02	-0.02	-0.07	0.02	-0.01
	(0.04)	(0.05)	(0.05)	(0.05)	(0.04)	(0.04)
Age 0-10 weeks	-0.04	-0.06	-0.06	-0.13**	0.01	0.02
	(0.06)	(0.06)	(0.06)	(0.06)	(0.06)	(0.07)
Age 11-22 weeks		-0.03		-0.05		-0.00
		(00.0)		(00.0)		(00.0)
Cohorts:	LO O					E C
Conceived after tornado	**/0.0- (20.02)	-0.02	c0.0-	-0.02	0.23***	0.1/***
	(0.03) 0.05 ±	(0.04) 0.04	(0.04) 0.05	(0.04)	(0.04)	(0.04) 0.11***
In utero	*0.05*	-0.04	c0.0-	-0.06*	0.02	0.11***
$\Delta co = 0.10$ mode	(cn.u) 0.0.0	(cn.u)	(cn.u)	(0.04) 0.04	(cn.u) 20.0	(cu.u) ***71 0
112C 0-10 WCCV3	20.0					0.04
A rea 11 22 mealse	(+0.0)	(+0.0)	(1-0-0)	0.11***	(+0.0)	(10.07) 0 13***
VAC 11-77 MCCKS		. / 0.0-				
		(0.04)		(0.04)		(0.04)
Post much land, of matrice (mode)	***700	***/00	0.05***	0.02***	0000	0000
Dest guess length of gestation (weeks)	0.00	0.04***	****CU.U	1.00 0	-0.00	-0.00
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Age at 3 monun measurement (centereu)	0.02 70.00		1.00 00 100 00		10.00 ox	
-	(00.0)	- 0 -	(00.0)	0.0004444	(0.00)	000
Age at 6 month measurement (centered)		0.01***		0.00***		0.00
		(0.00)		(0.00)		(0.00)
Observations	16490	16226	16471	16211	16776	16598
I incar recression models of anthronometric growth and fe	wer incidence using	- snecificat	ions similar	to our main	snerificati	nans excen
that we add a new cohort of those that were conceived after the	he fornado time (by	our hest a	mese) Hisin	or this snerifi	ration the c	omnarison

Observations1649016226164711677616598Linear regression models of anthropometric growth and fever incidence using specifications similar to our main specifications except that we add a new cohort of those that were conceived after the tornado time (by our best guess). Using this specification the comparison (reference) group includes only infants born more than 2.5 months before the tornado (5.5 months in even numbered columns). Standard errors are clustered at the sector level.Significance: * < 0.10, *** < 0.05; *** < 0.01.</td>

vo-way clustering
Ê
using
regressions
ey
\mathbf{M}
13:
Table

Triple int.: In tornado area X Vit A X Conceived after tornado	3 Mo. b/se	6 Mo.	3 Mo.	6 Mo.	3 Mo.	6 Mo.
Triple int.: In tornado area X Vit A X Conceived after tornado	b/se	1 1	1 /			
Triple int.: In tornado area X Vit A X Conceived after tornado		b/se	D/Se	b/se	b/se	b/se
Conceived after tornado						
	0.04	-0.13	-0.02	-0.21	-0.01	0.04
	(0.13)	(0.15)	(0.15)	(0.17)	(0.17)	(0.20)
In utero	0.20	0.11	0.21	0.19	0.04	0.03
0-10 weeks of age	(0.16) 0 42**	(c1.0) 0.47**	(0.17) (0.17) (0.43**	(0.17) 058***	(0.19) -0 47*	(0.17) -0.21
	(0.20)	(0.19)	(0.20)	(0.18)	(0.24)	(0.27)
11-22 weeks of age	~	0.06	~	0.19	~	-0.16
		(0.20)		(0.22)		(0.26)
Double interaction: In tornado area X						
Conceived after tornado	0.23^{***}	0.19*	0.22^{***}	0.14	-0.04	-0.13
	(0.08)	(0.11)	(0.0)	(0.13)	(0.11)	(0.15)
In utero	-0.09	-0.01	-0.11	-0.0- (01-0)	0.29***	0.11
	(11.0)	(0.10)	(0.11)	(0.12)	(0.11) 0.45**	(0.14)
U-10 WEEKS OI AGE	-0.24	-07.0-	(91.0)	-0.04 (0.10)	(81.0)	~/C.U
11-22 weeks of ace	(01.0)	(c1.0)	(01.0)	-0.20	(01.0)	(777.0)
22 MARY 10 MARY 11		(0.15)		(0.17)		(0.18)
Double interaction: Vit A X						
Conceived after tornado	-0.04	-0.06	-0.03	-0.06	0.02	0.02
	(0.04)	(0.05)	(0.05)	(0.05)	(0.05)	(0.05)
In utero	-0.00	-0.02	-0.02	-0.07	0.02	-0.01
	(0.04)	(0.05)	(0.05)	(0.05)	(0.04)	(0.04)
Age 0-10 weeks	-0.04	-0.06	-0.06	-0.13**	0.01	0.02
	(0.06)	(0.06)	(0.06)	(0.06)	(0.06)	(0.07)
Age 11-22 weeks		-0.03		-0.05		-0.00
		(006)		(00.0)		(006)
Consistent office to anothe	0 07**	000	0.05	000	***C U	0.17***
Concerved after tornado	-0.0/ /0.03/	70.04	CU.U-	70.04	0.40 M	0.1/0
In utero	(c0·0) *50 0-	(0.04) -0.04	(+0.0) -0.05	(0.04) -0.06*	(+0.0) 0.02	(0.04) 0 11***
	(0.03)	(0.03)	(0.03)	(0.04)	(0.03)	(0.03)
Age 0-10 weeks	0.02	0.04	0.04	0.04	0.05	0.17^{***}
	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
Age 11-22 weeks		-0.07*		-0.11***		0.13^{***}
		(0.04)		(0.04)		(0.04)
- Other controls - Best mess length of gestation (weeks)	0.06***	0.04***	0.05***	0.03***	-00	-0.00
	0000	0000			0000	0000
Age at 3 month measurement (centered)	0.02^{***}	(00.0)	0.02***	(00.0)	0.01***	(00.0)
0	(0.00)		(0.00)		(0.00)	
Age at 6 month measurement (centered)		0.01^{***}		0.00***		0.00
		(0.00)		(0.00)		(0.00)
Observations	16490	16226	16471	16211	16776	16598

tornado exposure.
ef.
levels (
Ξ
differen
0
y regressions t
S.
-
: Robustness of
4
Table 1

	IA		MU	IAC	Fev	er
	3 Mo. b/se	6 Mo. b/se	3 Mo. b/se	6 Mo. b/se	3 Mo. b/se	6 Mo. b/se
Triple: Tornado damage 0-20% X Vit A X						
In utero	0.24	0.12	0.17	0.15	-0.04	0.08
Arre 0-10 wreeks	(0.24) 0 33**	(0.17) 0.64***	(0.19) 0.35**	(0.18) 0 51***	(12.0)	(0.27)
	(0.17)	(0.17)	(0.17)	(0.18)	(0.42)	(0.54)
Age 11-22 weeks	~	0.18	~	0.42	~	-0.60
0		(0.35)		(0.36)		(0.43)
Triple: Tornado damage 20-100% X Vit A X	č			Ċ		0
In utero	0.20	0.24	0.32	0.42*	0.13	-0.06
Age 0-10 weeks	(0.21)	(0.20) 0.55**	(0.24) 0.55**	(0.22) 0.88***	(uc.u) 44.0-	(12.0) 0.09
	(0.24)	(0.25)	(0.24)	(0.23)	(0.34)	(0.33)
Age 11-22 weeks		0.07		0.28		0.03
Double interaction: Tornado damare (1.20% V		(0.21)		(0.25)		(0.38)
In utero	-0.33	-0.11	-0.25	-0.07	0.33^{**}	0.16
-	(0.22)	(0.12)	(0.15)	(0.14)	(0.16)	(0.23)
Age 0-10 weeks	-0.32^{***}	-0.56*** (0.10)	-0.34*** (0.06)	-0.46^{***}	0.18	0.53) (0.53)
Age 11-22 weeks		-0.13		-0.46		0.90**
		(0.28)		(0.30)		(0.36)
Double interaction: Iornado damage 20-100% A In utero	-0.09	-0.09	-0.16	-0.12	0.29**	0.17
	(0.10)	(0.10)	(0.13)	(0.12)	(0.13)	(0.13)
Age 0-10 weeks	-0.30	-0.26	-0.32	-0.37***	0.55***	0.38**
	(0.20)	(0.18)	(0.20)	(0.14)	(0.18)	(0.19)
Age 11-22 weeks		-0.16		-0.18 (0,17)		(0.23)
Double interaction: Vit A X		(
In utero	0.01	0.00	-0.01	-0.04	0.02	-0.02
-	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
Age U-1U Weeks	70.05	-0.04	CU.U-	-0.11*	10.0	10.0
Age 11-22 weeks	(00.0)	-0.00	(00.0)	-0.02	(00.0)	-0.01
) .		(0.05)		(0.05)		(0.05)
Conorts: In utero	-0.03	-0.03	-0.03	-0.06*	-0.05*	0.04
	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)
Age 0-10 weeks	0.05	0.05	0.06	0.04	-0.03	0.10 **
	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
Age 11-22 weeks		-0.06		-0.10***		0.0/*
- Other controls -		(+0.0)		(+0.0)		(+0.0)
Best guess length of gestation (weeks)	0.06^{***}	0.04^{***}	0.05***	0.03^{***}	-0.00	-0.00
•	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Age at $\dot{3}$ month measurement (centered)	0.02***		0.02***			
A as at 6 month measurement (centered)	(00.0)	0.01***	(00.0)	0 00***	(00.0)	-0.00
		(0.00)		(000)		(0.00)
Observations	16490	16226	16471	16211	16776	16598
Domotion models of low attended reading to the	to a do i o do	officeed in the	1/1		aton mb out	J 2007 0F

CONSETVATIONS 102-51 102-51 102-50 102-50 102-51 102-51 102-51 102-50 102-50 102-50 102-50 102-50 Regression models of key outcome variables where exposure to the tornado is defined in two ways: 1) being in a sector where 0.20% of houses were destroyed by the tornado. Standard errors are clustered at the (randomization) sector level. Significance: * < 0.10% sector level.

Triple interaction: In tornado area X Vit A X In utero	V	Г	MI	JAC	Fev	1
Triple interaction: In tornado area X Vit A X In utero						5
Triple interaction: In tornado area X Vit A X In utero	.0 IMI C	6 Mo.	3 Mo.	6 Mo.	3 Mo.	6 Mo.
Triple interaction: In tornado area X Vit A X In utero	b/se	b/se	b/se	b/se	b/se	b/se
In utero						
	0.25	0.17	0.22	0.22	-0.02	-0.00
	(0.17)	(0.13)	(0.16)	(0.14)	(0.19)	(0.18)
0-10 weeks of age	0.55**	0.75***	0.54 * *	0.80^{***}	-0.49*	-0.26
)	(0.23)	(0.18)	(0.23)	(0.18)	(0.25)	(0.30)
11-22 weeks of age		0.13		0.29		-0.37
		(0.20)		(0.22)		(0.30)
Double interaction: In tornado area X						
In utero	-0.22*	-0.09	-0.18	-0.05	0.37^{***}	0.18
	(0.13)	(0.0)	(0.11)	(0.0)	(0.13)	(0.14)
0-10 weeks of age	-0.46**	-0.57***	-0.43**	-0.53***	0.49^{**}	0.46*
	(0.20)	(0.13)	(0.20)	(0.13)	(0.20)	(0.25)
11-22 weeks of age		-0.16		-0.27		0.74^{***}
		(0.15)		(0.18)		(0.22)
Double interaction: Vit A X						
In utero	0.01	0.00	-0.01	-0.04	0.02	-0.02
	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
0-10 weeks of age	-0.02	-0.04	-0.05	-0.11*	0.01	0.01
)	(0.05)	(0.06)	(0.06)	(0.00)	(0.06)	(0.06)
11-22 weeks of age		-0.00		-0.02		-0.01
		(0.05)		(0.05)		(0.05)
Cohorts:						
In utero	-0.03	-0.03	-0.03	-0.06*	-0.05*	0.04
	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)
0-10 weeks of age	0.05	0.05	0.06	0.04	-0.03	0.10^{**}
	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
11-22 weeks of age		-0.06		-0.10***		0.07*
		(0.04)		(0.04)		(0.04)
- Outer controls - Dant amount of anteriors (amount)	***700	***100	***300	***CU U	000	000
Dest guess length of gestation (weeks)	0.00	0.04***	***CU.U	***0.0	00.0-	-0.00
•	(000)	(0.00)	(0.00)	(00.0)	(0.00)	(0.00)
Age at 3 month measurement (centered)	0.02^{***}		0.02^{***}		0.01^{***}	
	(0.00)		(0.00)		(0.00)	
Age at 6 month measurement (centered)		0.01^{***}		0.00^{***}		-0.00
		(0.00)		(0.00)		(0.00)
Observations	16332	16070	16313	16055	16617	16440

Table 15: Robustness of key regressions to excluding the most damaged sectors.

A Growth

Here we test whether growth, in addition to levels of anthropometric outcomes, was affected. To do this, we use growth in anthropometric measures (AI, MUAC, CC, and HC) from 0 to 3 and 0 to 6 months as the outcome variables in the same specification as worked with earlier. The results of these estimations are reported in Tables A2 and A3. Growth in both periods suffers for infants exposed to the tornado in early life. Vitamin A has a significant protective effect on growth measured at 3 and 6 months. Overall, consistent with the results on levels reported above, the evidence suggests that levels as well as growth are affected by tornado exposure in early life, but that the protective effect of vitamin A is manifest for both types of anthropometric measurement.

B Low Birth Weight Infants

In this section, we test whether babies with more general disadvantage at birth–in particular, low birth weight, as defined as birth weight below 2 kg–are worse off by 3 and 6 months, and whether vitamin A supplementation at birth is able to recuperate some of this disadvantage, in the same way that it does for tornado-exposed infants.⁷ Obviously, this analysis loses some of the advantages of using tornado exposure to indicate disadvantage in early life, because the tornado is plausibly exogenous. Birth weight, on the other hand, may be correlated with outcomes later in infancy for a variety of non-causal reasons. The advantage, however, is that the tornado is a very specific kind of shock, and not easily generalizable to a more general concept of disadvantage at birth, whereas birth weight is likely a good summary measure of and proxy for this concept.

To implement the analysis, we regress AI and fever episodes at 3 and 6 months on a dummy for birth weight less than 2 kg, a vitamin A treatment sector dummy, and the interaction of the two variables. We include the same controls as we do in our baseline specification.

Results are reported in Table A4. As expected, babies with low birth weight have significantly worse AI by 3 and 6 months—about 1 standard deviation lower than "normal" infants. They are also more likely to have had episodes of serious fevers in infancy. The interaction with vitamin A treatment is small and insignificantly different from 0 for AI, indicating that there is no significant mitigating impact

⁷We chose a cutoff of 2 kg rather than the internationally used 2.5 kg because, given the very low-income study area, nearly all the infants in our sample are below 2.5 kg so the 2.5 kg partition does not generate much variation.

of vitamin A supplementation on outcomes at 3 and 6 months. On the other hand, if we look to fever episodes, by 6 months the mitigative impact of vitamin A is quite clear (column 4).

Taking this evidence together with our main results on resilience to the negative effects of a tornado, then, we see that vitamin A may not generate resilience for all types of disadvantage, but rather may protect against the negative impacts of certain types of trauma. In particular, given the mechanisms through which vitamin A acts – that is, strengthening infants' immune systems – it seems plausible that types of trauma that test or weaken the immune system are those against which vitamin A might exhibit the largest protective effects.

C Appendix Tables

area
tornado
the
outside
and
nside
cohort i
by
Attrition
A1:
Table .

			Ir	l torna	do are	a					°Z	t in to	rnado a	rea		
	Pre-		0-24		In-		Post-		Pre-		0-24		In-		Post-	
	torna	op	week	s	utero		torn	opı	torna	op	weeks		utero		tornad	0
	Z	%	Ζ	%	Z	%	Ζ	%	Z	%	Z	%	Z	%	Z	%
Singleton births in the infant trial	209		315		340		266		3003		4331		4520		3682	
Non missing vital status at 24 weeks	208	100	315	100	339	100	266	100	2998	100	4325	100	4514	100	3682	100
At birth																
Non missing birth anthropometry	195	93	305	70	322	95	253	95	2756	92	4058	94	4289	95	3461	94
Non missing or late birth anthr.	166	79	260	83	276	81	220	83	2322	77	3409	79	3622	80	3002	82
At 3 months																
Alive at 3 months	198	95	302	96	325	96	253	95	2839	95	4084	94	4331	96	3434	93
Non missing 3 month anthropometry	191	91	290	92	306	90	241	91	2632	88	3825	88	4088	90	3288	89
Non missing or late 3 month anthr.	189	90	284	90	301	89	236	89	2597	86	3791	88	4029	89	3264	89
At 6 months																
Alive at 6 months	197	94	301	96	320	94	252	95	2819	94	4063	94	4313	95	3416	93
Non missing 6 month anthropometry	195	93	291	92	305	90	235	88	2553	85	3749	87	4015	89	3202	87
Non missing or late 6 month anthr.	193	92	289	92	304	89	235	88	2495	83	3699	85	3987	88	3194	87
The table lists the number of infants by cohort and a that fulfill the given data requirements (e.g., having n	area that we	re born anthroi	during	the tri ic data	al in th at birtl	e first 1 1 (Row	row. In 7 3)). Ir	subsequ analyse	lent rows s we use	it lists : only sin	number ngleton l	of infa	nts and J this sam	ple is g	age of th iven in F	tow 2).

first 24 weeks of life (Column 2) or in-utero (Column 3). Infants in the post-tornado cohort (Column 4) are those conceived after the tornado (based on reported last menstrual period). In Column 5 (Unknown) are infants for which we do not have enough data to determine the cohort (these are excluded from the analysis). Anthropometry at birth is considered on time if taken no later than 7 days after birth. Anthropometry at 3 and 6 months is considered on time if taken no later than 6 weeks after the target date (target date is 12 and 24 weeks, respectively). Pre-tornado (Column 1) is the cohort of infants that were at least 6 months of age when the tornado hit. The next two columns are the cohorts affected by the tornado, in the

		II	Δ M	UAC	0	CC	ΔF	IC
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se
Triple interaction: In tornado area X Vit A X								
In utero	0.21	0.20	0.18	0.18	0.38	0.38	0.21	0.20
	(0.13)	(0.13)	(0.16)	(0.15)	(0.29)	(0.29)	(0.16)	(0.17)
0-10 weeks of age	0.34^{**}	0.30*	0.34^{**}	0.29*	0.77*	0.72*	0.19	0.17
	(0.17)	(0.17)	(0.16)	(0.16)	(0.41)	(0.41)	(0.20)	(0.19)
Double interaction: In tornado area X								
In utero	-0.04	-0.03	-0.03	-0.02	-0.23	-0.20	0.04	0.05
	(0.10)	(0.10)	(0.12)	(0.11)	(0.21)	(0.20)	(0.13)	(0.13)
0-10 weeks of age	-0.32***	-0.31***	-0.31***	-0.30***	-0.90***	-0.89***	-0.08	-0.08
	(0.11)	(0.11)	(0.11)	(0.11)	(0.29)	(0.30)	(0.14)	(0.13)
Double interaction: Vit A X								
In utero	-0.09**	-0.08**	-0.08**	-0.07*	-0.14*	-0.12	-0.10**	-0.09**
	(0.04)	(0.04)	(0.04)	(0.04)	(0.08)	(0.08)	(0.04)	(0.04)
0-10 weeks of age	-0.10*	-0.09*	-0.10*	-0.09	-0.21*	-0.21*	-0.06	-0.06
	(0.06)	(0.06)	(0.06)	(0.06)	(0.12)	(0.12)	(0.08)	(0.08)
Cohorts:								
In utero	0.03	0.01	0.01	-0.01	0.10*	0.07	0.00	-0.01
	(0.03)	(0.03)	(0.03)	(0.03)	(0.06) 	(0.06)	(0.03)	(0.03)
0-10 weeks of age	0.09**	0.08**	0.07*	0.07	0.11	0.10	0.10*	0.10*
	(0.04)	(0.04)	(0.04)	(0.04)	(0.08)	(0.08)	(cn.0)	(cn.n)
- Other controls - Best guess length of gestation (weeks)								
Age at 3 month measurement (centered)	0.03 * * *	0.03***	0.02***	0.02***	0.04^{***}	0.04 * * *	0.03 * * *	0.03 * * *
Weight at hirth (kg)	(00.0) 0 57***	(00.00) 0 53***	(00.0) 0.45***	(0.00) 0.41***	1 00***	(00.0) 0 03***	(0.00) 0.61***	(00.00) 0 58***
weight at Diffic (Ng)	(JO 05)	(0.05)	(20.02)	0.000	0.10	0100	10.06	0.06
Height at birth (cm)	(cn·n)	0.05***	(0.0)	0.03***	0.12^{***}	$0.10 \\ 0.11^{***}$	(00.0) 0.07***	(00.0) 0.07***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
MUAC at birth (cm)	-0.38***	-0.39***	-0.77***	-0.77***	-0.13***	-0.14***	-0.15***	-0.15***
~	(0.02)	(0.02)	(0.02)	(0.02)	(0.04)	(0.04)	(0.02)	(0.02)
Head circumference at birth (cm)	-0.13***	-0.13***	0.06***	0.05***	0.10^{***}	0.10^{***}	-0.56***	-0.57***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.02)
Chest circumterence at birth (cm)	-0.12***	-0.13***	0.02**	0.02**	-0.68***	-0.69***	-0.00	0.00-
Maternal MIIAC	(10.0)	(10.0) 0.04***	(10.0)	(10.0) 0.05***	(20.0)	(20.0) 0.07***	(10.0)	(10.0) 0.02***
		(00.0)		(00.0)		(0.01)		00 00)
Maternal height		0.01***		0.01***		0.03***		0.01^{***}
O		(0.00)		(0.00)		(0.00)		(0.00)
Living Standards Index		0.08^{***}		0.08***		0.16^{***}		0.05***
0		(0.01)		(0.01)		(0.02)		(0.01)
Observations	12279	12260	12268	12249	11917	11901	12252	12233
		J.	-1.L.					
Linear regression models of anturopointeur grown at 3 month birth and 3 months (Δ MUAC = MUAC at 3 months - MUAC at variables (Δ MUAC, Δ CC and Δ HC) after each has been standa	t birth) measur t birth) measur ardized to zero	red in centin mean and u	auous. 1 ne neters. ΔAI nit standard	is a standar deviation. I	dized (zero	mean, unit S mean, unit S ards is an inc	D) average or the based or	of the three of a principal
components analysis of household assets. Standard errors are clu	stered at the se	ector level.)			
Significance: $* < 0.10$; $** < 0.05$; $*** < 0.01$.								

Table A2: Anthropometry Growth at 3 Months.

at 6 Months.
Growth
Anthropometry
A3:
Table

Triple interaction: In tornado area X Vit A X In utero	n/se	n/ se	n/ sc	n/se	n/ se	n/ se	n/sc	n/ se
Triple interaction: In tornado area X Vit A X In utero								
11 III ero	5 5 5	÷.00				, c c c	L 7 0	
	(0.13)	(0.12)	0.28° (0.16)	(0.15)	(0.31)	0.29) (0.29)	c1.0 (0.19)	(0.19)
0-10 weeks of age	0.43**	0.39**	0.65***	0.59***	0.39	0.32	0.46*	0.44*
11-22 weaks of are	(0.18)	(0.17)	(0.20) 0.29	(0.19) 0.28	(0.47)	(0.45)	(0.23)	(0.23)
11-27 MCCNS OL 980	(0.23)	(0.24)	(0.23)	(0.25)	(0.51)	(0.52)	(0.26)	(0.26)
Double interaction: In tornado area X	0.02		10.0	0.01	0.05	0.01	000	20.0
	c0:0- (60:0)	-0.05	-0.01	0.11)	-0.05 (0.23)	-0.01	-0.03	-0.07
0-10 weeks of age	-0.32***	-0.31***	-0.40***	-0.38***	-0.54	-0.52	-0.29	-0.28*
	(0.12)	(0.11)	(0.11)	(0.10)	(0.36)	(0.33)	(0.19)	(0.17)
11-22 weeks of age	-0.31^{**}	-0.30^{**}	-0.5/**	-0.36^{*}	-0.6/** (0.30)	-0.64^{**} (0.31)	-0.18 (0.15)	-0.17 (0.16)
Double interaction: Vit A X	~	~	~	~	~	~	~	~
In utero	-0.08*	-0.07	-0.09**	-0.08*	-0.11	-0.09	-0.08	-0.08
0-10 weeks of age	(0.04) -0.09	-0.08 -0.08	(cn.u) -0.13**	(cu.u) -0.12**	-0.18 -0.18	(20.0) -0.18	(00.0) -0.05	(0.00) -0.04
	(0.06)	(0.06)	(0.06)	(0.06)	(0.13)	(0.13)	(0.09)	(0.09)
11-22 weeks of age	0.04	0.04	0.02	0.02	0.05	0.05	0.08	0.08
Cohorts:	(00.0)	(co.o)	(00.0)	(00.0)	(0.14)	(11:0)	(10.0)	(10.0)
In utero	0.02	0.00	-0.01	-0.03	0.10	0.06	0.03	0.02
0-10 weeks of age	(0.03)	(0.03) 0.09**	(0.03) 0.08*	(0.03) 0.07*	(0.07) 0.18*	(0.07)	(0.04)	(0.04)
	(0.04)	(0.04)	(0.04)	(0.04)	(0.09)	(0.09)	(0.06)	(0.06)
11-22 weeks of age	0.03	0.02	-0.02	-0.03	0.01	000	0.12**	0.12**
- Other controls - Best guess length of gestation (weeks)		(10.0)	(10.0)	(+0.0)	(00.0)	(00.0)	(00.0)	(60.0)
Age at 6 month measurement (centered)	0.01^{***}	0.01^{***}	0.00***	0.00***	0.02^{***}	0.02***	0.02^{***}	0.02***
Weicht at hirth (ko)	(0.00) 0.48***	(0.00) 0.43***	(0.00) 0.43***	(0.00) 0.38***	(0.0)	(0.00)	(0.00) 0.57***	(0.00) 0.52***
	(0.05)	(0.05)	(0.05)	(0.05)	(0.11)	(0.11)	(0.07)	(0.07)
Height at birth (cm)	0.05***	0.04***	0.03***	0.03***	0.10***	0.09***	0.06***	0.06***
MUAC at birth (cm)	-0.39***	(0.01) -0.40***	(10.0) ***67.0-	(10.0) -0.80***	(0.01) -0.23***	(0.01) -0.24***	(10.01) -0.17***	(10.0) -0.17***
Ilond riserum freenen of histly (rem)	(0.02) 0.14***	(0.02) 0.15***	(0.02)	(0.02)	(0.04)	(0.04)	(0.02) 0 50***	(0.02) 0 50***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Chest circumference at birth (cm)	-0.14***	-0.15***	-0.01	-0.01	-0.75***	-0.76***	-0.04***	-0.04***
Maternal MUAC	(10.0)	(0.01) 0.05***	(10.0)	(10.0) ****90.0	(20.0)	(70.0) ***60.0	(10.0)	(10.0) 0.03***
		(0.00)		(0.00)		(0.01)		(0.01)
Maternal height		0.00 00.00		0.01***		0.03***		0.02***
Living Standards Index		(000) ***60.0		0.11^{***}		0.20***		(00.0) ***90.0
		(0.01)		(0.01)		(0.02)		(0.01)
Observations	12016	11996	12010	11990	11554	11535	11938	11918

	Α	I	Fever ej	pisodes
	3 months	6 months	3 months	6 months
	b/se	b/se	b/se	b/se
Vitamin A X Birth weight < 2kg	0.03	0.00	-0.05	-0.12**
5	(0.05)	(0.05)	(0.06)	(0.05)
Vitamin A	0.01	0.01	0.01	0.00
	(0.02)	(0.02)	(0.02)	(0.02)
Birth weight < 2kg	-1.19***	-0.93***	0.09**	0.11^{***}
)	(0.04)	(0.04)	(0.04)	(0.04)
Age in days at 3 month visit	0.02^{***}		0.00***	
	(0.00)		(0.00)	
Age in days at 6 month visit		0.01^{***}		-0.00
		(0.00)		(0.00)
Constant	-1.78***	-1.41***	0.45^{***}	1.01^{***}
	(0.07)	(0.16)	(0.08)	(0.16)
Observations	14955	14631	15156	14913
Linear regression models of infant outcomes by birth weight. The c at 3 and 6 months (standardized average of MUAC, CC and HC) and coded at 4). All regressions include sector (randomization unit) fixe	outcome varia the number of ed effects. Sta	bles are the A of fever episo indard errors	Anthropometr des at 3 and 6 are clustered	ic Index (AI) months (top at the sector

Table A4: Vitamin Supplementation for Low Birthweight Infants

level. Significance: * < 0.10; ** < 0.05; *** < 0.01.

			Dosed at		
	$\leq = 6$ hours	$\leq = 12$ hours	<= 18 hours	$\leq = 24$ hours	$\leq = 7 \text{ days}$
	b/se	b/se	b/se	b/se	b/se
Double interaction: In tornado area X					
In utero	-0.05	-0.03	-0.02	-0.00	-0.01
	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)
0-10 weeks of age	-0.00	0.00	0.05	0.05	0.02
	(0.06)	(0.06)	(0.05)	(0.05)	(0.03)
Cohorts:					
In utero	0.00	0.01	-0.00	-0.01	-0.01
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
0-10 weeks of age	-0.05***	-0.05***	-0.04***	-0.03***	-0.01
1	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Constant	0.47 * * *	0.63^{***}	0.71^{***}	0.75***	0.86 * * *
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Observations	16812	16812	16812	16812	16812
I incompared to the second of the second	Constituent Standard	oracia cao clinoto	and of the conten]	[]	

Table A5: Time at dosing.

Linear regression models of time at dosing using our main specifications. Standard errors are clustered at the sector level. Significance: * < 0.10, ** < 0.05, *** < 0.01.